

Intensity of continuous renal replacement therapy for acute kidney injury (Review)

Fayad AI, Buamscha DG, Ciapponi A

Fayad AI, Buamscha DG, Ciapponi A. Intensity of continuous renal replacement therapy for acute kidney injury. *Cochrane Database of Systematic Reviews* 2016, Issue 10. Art. No.: CD010613. DOI: 10.1002/14651858.CD010613.pub2.

www.cochranelibrary.com



TABLE OF CONTENTS

[Intervention Review]

Intensity of continuous renal replacement therapy for acute kidney injury

Alicia I Fayad¹, Daniel G Buamscha², Agustín Ciapponi³

¹Pediatric Nephrology, Ricardo Gutierrez Children's Hospital, Buenos Aires, Argentina. ²Pediatric Critical Care Unit, Juan Garrahan Children's Hospital, Buenos Aires, Argentina. ³Argentine Cochrane Centre, Institute for Clinical Effectiveness and Health Policy (IECS-CONICET), Buenos Aires, Argentina

Contact address: Alicia I Fayad, Pediatric Nephrology, Ricardo Gutierrez Children's Hospital, Institute for Clinical Effectiveness and Health Policy, Los Incas Av 4174, Buenos Aires, 1427, Argentina. aliciafayad@gmail.com, afayad@fibertel.com.ar.

Editorial group: Cochrane Kidney and Transplant Group. **Publication status and date:** New, published in Issue 10, 2016. **Review content assessed as up-to-date:** 9 February 2016.

Citation: Fayad AI, Buamscha DG, Ciapponi A. Intensity of continuous renal replacement therapy for acute kidney injury. *Cochrane Database of Systematic Reviews* 2016, Issue 10. Art. No.: CD010613. DOI: 10.1002/14651858.CD010613.pub2.

Copyright © 2016 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Acute kidney injury (AKI) is a common condition among patients in intensive care units (ICU), and is associated with substantial morbidity and mortality. Continuous renal replacement therapy (CRRT) is a blood purification technique used to treat the most severe forms of AKI but its effectiveness remains unclear.

Objectives

To assess the effects of different intensities (intensive and less intensive) of CRRT on mortality and recovery of kidney function in critically ill AKI patients.

Search methods

We searched Cochrane Kidney and Transplant's Specialised Register to 9 February 2016 through contact with the Information Specialist using search terms relevant to this review. Studies contained in the Specialised Register are identified through search strategies specifically designed for CENTRAL, MEDLINE, and EMBASE; handsearching conference proceedings; and searching the International Clinical Trials Register (ICTRP) Search Portal and ClinicalTrials.gov. We also searched LILACS to 9 February 2016.

Selection criteria

We included all randomised controlled trials (RCTs). We included all patients with AKI in ICU regardless of age, comparing intensive (usually a prescribed dose \geq 35 mL/kg/h) versus less intensive CRRT (usually a prescribed dose < 35 mL/kg/h). For safety and cost outcomes we planned to include cohort studies and non-RCTs.

Data collection and analysis

Data were extracted independently by two authors. The random-effects model was used and results were reported as risk ratios (RR) for dichotomous outcomes and mean differences (MD) for continuous outcomes, with 95% confidence intervals (CI).

Main results

We included six studies enrolling 3185 participants. Studies were assessed as being at low or unclear risk of bias. There was no significant difference between intensive versus less intensive CRRT on mortality risk at day 30 (5 studies, 2402 participants: RR 0.88, 95% CI 0.71 to 1.08; $I^2 = 75\%$; *low quality of evidence*) or after 30 days post randomisation (5 studies, 2759 participants: RR 0.92, 95% CI 0.80 to 1.06; $I^2 = 65\%$; *low quality of evidence*). There were no significant differences between intensive versus less intensive CRRT in the numbers of patients who were free of RRT after CRRT discontinuation (5 studies, 2402 participants: RR 1.12, 95% CI 0.91 to 1.37; $I^2 = 71\%$; *low quality of evidence*) or among survivors at day 30 (5 studies, 1415 participants: RR 1.03, 95% CI 0.96 to 1.11; I $^2 = 69\%$; *low quality of evidence*) and day 90 (3 studies, 988 participants: RR 0.98, IC 95% 0.94 to 1.01, $I^2 = 0\%$; moderate*quality of evidence*). There were no significant differences between intensive CRRT on the number of days in hospital (2 studies, 1665 participants): MD -0.23 days, 95% CI -3.73 to 2.89; $I^2 = 8\%$; *low quality of evidence*) and the number of days in ICU (2 studies, 1665 participants: MD -0.58 days, 95% CI -3.73 to 2.56, $I^2 = 19\%$; *low quality evidence*) compared to less intensive CRRT increased the risk of hypophosphataemia (1 study, 1441 participants: RR 1.21, 95% CI 1.11 to 1.31; *high quality evidence*) compared to less intensive CRRT. There was no significant differences between intensive and less intensive CRRT on numbers of patients who experienced adverse events (3 studies, 1753 participants: RR 1.08, 95% CI 0.73 to 1.61; $I^2 = 16\%$; *moderate quality of evidence*). In the subgroups analysis by severity of illness and by aetiology of AKI, intensive CRRT would seem to reduce the risk mortality (2 studies, 531 participants: RR 0.73, 95% CI 0.61 to 0.88; $I^2 = 0\%$; *high quality of evidence*) only in the subgroup of patients with post-surgical AKI.

Authors' conclusions

Based on the current low quality of evidence identified, more intensive CRRT did not demonstrate beneficial effects on mortality or recovery of kidney function in critically ill patients with AKI. There was an increased risk of hypophosphataemia with more intense CRRT. Intensive CRRT reduced the risk of mortality in patients with post-surgical AKI.

PLAIN LANGUAGE SUMMARY

Intensity of continuous renal replacement therapy for acute kidney injury

What is the issue?

Acute kidney injury (AKI) is very common among patients admitted to intensive care units (ICU), it is associated with a high death rated and characterised by the rapid loss of the kidney function. Patients with AKI show increased levels of serum uraemic toxins (creatinine and urea), serum potassium and metabolic acids, accumulation of water and in the most cases a reduction in urine output. In this population these chemicals and fluid overload are related to increased rates of death. Theoretically, effective removal of toxins and excess water from the bloodstream might improve patient outcomes (such as mortality rate and recovery of kidney function).

Continuous renal replacement therapy (CRRT) is a blood purification technique that enables removal of excess water and toxins. CRRT involves blood being diverted from the patient via a catheter (a hollow, flexible tube placed into a vein) through a filtering system which continuously and steadily removes excess water and toxins; purified blood is then returned to the patient via the catheter. Higher intensity CRRT improves the removal of toxins and excess water. The aim of this review was to investigate the effect of different intensities of CRRT (intensive or less intensive) on death, recovery of kidney function, and adverse events in people with AKI who are critically ill.

What did we do?

We searched the literature up until February 2016 and identified six studies enrolling 3185 patients with AKI that were evaluated in this review.

What did we find?

Six randomised studies enrolling 3185 participants were included in our review. Compared to less intensive CRRT, intensive CRRT did not reduce the risk of death, improve the recovery of kidney function, or reduce the risk of adverse events (such as bleeding) in patients with AKI. Intensive CRRT was associated with an increased risk of low blood phosphate levels.

SUMMARY OF FINDINGS FOR THE MAIN COMPARISON [Explanation]

Intensive versus less intensive CRRT for AKI

Patient or population: patients with AKI

Settings: ICU

Intervention: Intensive versus less intensive CRRT

| intervention. Intensive | intervention. Intensive versus less intensive on int | | | | |
|--------------------------------------------------------------------------|------------------------------------------------------|-------------------------------------|-------------------------------|----------------------------------|------------------------------------------------------|
| Outcomes | Illustrative comparative risks* (95% CI) | | Relative effect (95% Cl) | No. of participants (studies) | Quality of the evidence Comments (GRADE) |
| | Assumed risk | Corresponding risk | | | |
| | Less intensive CRRT | Intensive CRRT | | | |
| Mortality at day 30 | Study population | | RR 0.88 | 2402 (5) | $\Phi\Phi \odot \odot$ |
| Follow-up: 30 days | 430 per 1000 | 420 per 1000 (412 to 523) | (0.81 to 1.1) | | 10W ^{1,2} |
| | Moderate | | | | |
| | | | | | |
| Mortality after 30 days post-randomisation Follow-up: 60 days | Study population | | RR 0.92 | 2759 (5) | |
| | 514 per 1000 | 483 per 1000 (416 to 565) | (0.80 to 1.06) | | |
| | Moderate | | _ | | |
| | 593 per 1000 | 557 per 1000 (480 to 652) | | | |
| Patients free of RRT af- ter discontinuing CRRT Follow-up: 30 days | Study population | | RR 1.12 (0.91 to 1.37) | 2402 (5) | $\oplus \oplus \bigcirc \bigcirc$ low ^{1,2} |

ω

| | 483 per 1000 | 541 per 1000 (439 to 661) | | | |
|---------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|---------------------------------------------------------------|--------------------------------------------|-----------------------------------------------|
| | Moderate | | | | |
| | 390 per 1000 | 437 per 1000 (355 to 534) | | | |
| Patients free of RRT af- | Study population | | RR 0.98 | 988 (3) | ⊕⊕⊕⊖ moderate ³ |
| Follow-up: 90 days | 923 per 1000 | 904 per 1000 (867 to 932) | (0.94 to 1.01) | | |
| | Moderate | | | | |
| | 800 per 1000 | 784 per 1000 (752 to 808) | | | |
| Adverse events: hy- | Study population | | RR 1.21 | 1441 (1) | |
| pophosphataemia | 540 per 1000 | 654 per 1000 (600 to 708) | (1.11 to 1.31) | | nığı |
| | Moderate | | | | |
| | 540 per 1000 | 653 per 1000 (599 to 707) | | | |
| *The basis for the assu based on the assumed r CI: Confidence interval/ | m ed risk (e.g. the n isk in the compariso s; RR: Risk ratio; RR1 | nedian control group risk n group and the relative e T: renal replacement thera | across studies) is provi effect of the intervention apy | ded in footnotes. The (and its 95% CI). | corresponding risk (and its 95% confidence in |

Moderate quality: Further research is likely to have an important impact on our confidence in the estimate of effect and may change the estimate. Low quality: Further research is very likely to have an important impact on our confidence in the estimate of effect and is likely to change the estimate. Very low quality: We are very uncertain about the estimate.

4

² imprecision: due to wide Cl which crossed the threshold for clinically meaningful effects
 ³ Indirectness: critically ill patients with AKI in CRRT have high short-term mortality risk; mortality is a competing end point

for kidney recovery at day 90

ы

BACKGROUND

Description of the condition

Acute kidney injury (AKI) is a complex clinical entity characterised by an abrupt decline in kidney function (Mehta 2007). AKI incidence among adults admitted to intensive care units (ICU) range from 5% to 20% (Hoste 2006; Joannidis 2005); in children the incidence is 10% (Schneider 2010). Despite its potential to be reversed, AKI is associated with high rates of morbidity and mortality (Bagshaw 2007). AKI-related mortality substantially increases among people with multi-organ failure, sepsis or who are receiving renal replacement therapy (RRT) (Metnitz 2002; Sutherland 2010). More than 70% of people with AKI need renal support therapies. Despite advances in clinical care, people with AKI are at high risk of mortality and morbidity, and require significant healthcare resources (Sutherland 2010; Uchino 2005).

Description of the intervention

Continuous renal replacement therapy (CRRT) is an extracorporeal blood purification therapy, intended to support impaired kidney function. CRRT slowly removes fluid over prolonged periods (Foland 2004; Gibney 2008; Goldstein 2001; Mehta 1999), removes higher molecular weight solutes efficiently (Brunnet 1999; Clark 1999; Liao 2003; Ronco 2002; Sieberth 1995), and confers beneficial haemodynamic stability effects. CRRT modalities are defined by their main solute clearance mechanism. These are convection (continuous venovenous haemofiltration (CVVH)), diffusion (continuous venovenous haemodialysis (CVVHD)), or a combination of both convection and diffusion (continuous venovenous haemodiafiltration, CVVHDF) (Palevsky 2002). Several interventions have been used over the past three decades with the aim of improving poor prognoses of people with AKI. A significant factor that may impact on CRRT outcomes is intensity of treatment (timing of CRRT for AKI is being investigated in another Cochrane review, Fayad 2013a).

CRRT intensity is generally related to the quantity of solute removal required to improve outcomes in people with AKI. CRRT intensity can be analysed based either on solute removal from the blood, or appearance of solutes in effluent fluid. Some published studies have used effluent flow rates, expressed as total effluent volume/weight and unit of time (mL/kg/h), as a dose surrogate (RENAL Study 2006; Ronco 2000a), while accounting for effects of pre-dilution and modality differences (Claure-Del Granado 2011). Elsewhere, authors have considered that dialysis doses delivered as total effluent volume/clearance of solutes such as urea, creatinine is a better method to measure dose (Lyndon 2012). Equivalent renal urea clearance also provides a good estimate of delivered dialysis dose in CRRT (Claure-Del Granado 2012) which can be converted to effluent rate and expressed as mL/kg/h (Marshall 2006).

Few studies have assessed other dimensions of dose such as electrolyte and acid-base homeostasis (Bellomo 2013; Bihorac 2005; Morimatsu 2003; Uchino 2001) and fluid balance/fluid overload (Bouchard 2009; Davenport 2010; Sutherland 2010) using effluent volume as the dose measure.

How the intervention might work

A hypothesis that high intensity of RRT may improve survival has emerged from animal and human studies. These findings include indirect evidence from patients with ESKD (Lowrie 1981; Parker 1994).

Intensity based on a urea kinetics model was evaluated in animal studies by Grootendorst 1992, and in severe ill patients (sepsis, sepsis-shock) who received high dose (60 to 80 mL/kg/h) reported improvement in haemodynamic state with possible benefits in clinical outcomes (Honore 2000). A retrospective study found that dose correlated with survival in patients with intermediate scores of illness (Paganini 1996). Although prospective dose studies demonstrated association of improved survival or kidney recovery with high dose dialysis (Phu 2002; Ronco 2000a; Saudan 2006), these advantages were not universally observed (ATN Study 2005; Negash 2011; RENAL Study 2006; Vesconi 2009; Van Wert 2010).

Few studies have researched other components of dose that play important roles in clinical results. These include fluid balance and fluid overload associated with increased mortality risk (Bouchard 2009; Goldstein 2001), adequate homeostasis of electrolytes (sodium, potassium and hydrogen ions) related to cardiovascular stability, and the maintenance of kidney blood flow (Uchino 2001).

Why it is important to do this review

Studies assessing CRRT intensity (intensive versus less intensive) have either not reported investigation of all variables inherent in therapy for people with AKI or report inconsistent results. We investigated the relationship between different intensities of CRRT and clinical outcomes for people with AKI. Review evidence could have direct relevance to decisions about optimal intensity of CRRT to improve survival in critically ill patients with AKI.

OBJECTIVES

To assess the effects of different intensities (intensive and less intensive) of CRRT on mortality and recovery of kidney function in critically ill AKI patients.

METHODS

Criteria for considering studies for this review

Types of studies

All RCTs and quasi-RCTs (RCTs in which allocation to treatment was obtained by alternation, use of alternate medical records, date of birth or other predictable methods) looking at CRRT modalities for people with AKI in ICU settings were eligible for inclusion. For outcomes such us safety and costs, non-RCTs and cohort studies were also to be included if sufficiently high quality, sampling was clearly described, patients characterised, proportions of patients experiencing any adverse events or who dropped out because of adverse events was adequately reported, co-interventions were described, and at least 80% of patients included were analysed after treatment.

Types of participants

Inclusion criteria

We included all patients with AKI in ICU being treated with CRRT regardless of age or gender. We assigned AKI definitions cited by the included studies.

Exclusion criteria

We excluded patients who received dialysis treatment before admission to ICU, patients admitted for drug overdose (doses exceeding therapeutic requirements), or with acute poisoning (all toxins).

Types of interventions

We compared intensive (usually a prescribed dose \geq 35 mL/kg/h) versus less intensive CRRT (usually a prescribed dose < 35 mL/kg/h). These categories of intensities were defined as published in the original publications. We included all CRRT modalities (CVVH, CVVHD and CVVHDF).

Types of outcome measures

Primary outcomes

Death

- Death from any cause at days 7, 15, 30, 60, and 90
- Death or non-recovery at 90 days.

Recovery of kidney function

• Numbers of patients free of RRT after discontinuing CRRT

• Numbers of patients free of RRT after discontinuing CRRT at days 30, 60, and 90.

Secondary outcomes

Metabolic balance

• Numbers of patients who normalised serum electrolytes (potassium, sodium) concentration during CRRT

• Numbers of patients who normalised serum bicarbonate and base-excess concentration during CRRT

• Numbers of patients who normalised serum urea and creatinine concentration during CRRT.

Fluid balance

• Numbers of patients who achieved adequate fluid balance during CRRT.

Adverse events

• Numbers of patients who dropped out because of adverse events (technique or patient-dependent factors)

- Numbers of patients experiencing any adverse events
- Numbers of patients with intervention-related

complications (e.g. disequilibrium, hypokalaemia, hypophosphataemia, hypocalcaemia, bleeding, hypotension)

• Numbers of patients with catheter-related complications (early and late).

We looked for differences in overall dropout rates and any adverse effects by type (mild or severe). We defined adverse events severity where medical therapeutic interventions were implied in reporting. Withdrawals due to protocol violation or loss to follow-up were not included in counts of adverse events.

Length of stay

- Days in hospital
- Days in ICU.

Cost

We planned to assess costs of CRRT modalities including:

- Type and number of dialyser filters
- Use/no use of anticoagulation
- Types of anticoagulation and anticoagulants
- Use of replacement fluid
- Numbers of days on CRRT.

All costs were to be reported in international monetary units.

• Cost per day of CRRT (expressed in international monetary units)

- Length of hospital stay with CRRT
- Length of ICU stay with CRRT.

Search methods for identification of studies

Electronic searches

We searched the Cochrane Kidney and Transplant Specialised Register to 9 February 2016 through contact with the Information Specialist using search terms relevant to this review. The Specialised Register contains studies identified from the following sources.

1. Quarterly searches of the Cochrane Central Register of Controlled Trials (CENTRAL)

2. Weekly searches of MEDLINE OVID SP

3. Handsearching of kidney-related journals and the proceedings of major kidney conferences

- 4. Searching of the current year of EMBASE OVID SP
- 5. Weekly current awareness alerts for selected kidney journals

6. Searches of the International Clinical Trials Register

(ICTRP) Search Portal and ClinicalTrials.gov.

Studies contained in the Specialised Register were identified through search strategies for CENTRAL, MEDLINE and EM-BASE based on the scope of Cochrane Kidney and Transplant. Details of these strategies as well as a list of handsearched journals, conference, proceedings and currents awareness alert, available in the Specialised Register section of information about Cochrane Kidney and Transplant.

See Appendix 1 for search terms used in strategies for this review.

Searching other resources

1. LILACS (Latin American and Caribbean Health Sciences) (from March 1980 to February 2016)

2. Reference lists of review articles, relevant studies and clinical practice guidelines.

3. Letters seeking information about unpublished or incomplete studies to investigators known to be involved in previous studies.

Data collection and analysis

Selection of studies

The search strategy described was used to obtain titles and abstracts of studies with potential relevance to the review. Titles and abstracts were screened independently by two authors who discarded studies that were not applicable; however studies and reviews that could include relevant data or information on studies were retained initially. Two authors independently assessed retrieved abstracts, and if necessary, the full text of these studies to determine which satisfied the inclusion criteria.

Data extraction and management

Data extraction was carried out independently by two authors using standard data extraction forms. Studies reported in non-English language journals were translated before assessment. Where more than one publication of one study existed, reports were grouped together and the publication with the most complete data was used in the analyses. Where relevant outcomes were only published in earlier versions these data were used. We resolved any discrepancy by discussion.

Assessment of risk of bias in included studies

The following items were independently assessed using the risk of bias assessment tool (Higgins 2011) (see Appendix 2).

- Was there adequate sequence generation (selection bias)?
- Was allocation adequately concealed (selection bias)?
- Was knowledge of the allocated interventions adequately prevented during the study?
 - Participants and personnel (performance bias)
 - Outcome assessors (detection bias)

• Were incomplete outcome data adequately addressed (attrition bias)?

• Are reports of the study free of suggestion of selective outcome reporting (reporting bias)?

• Was the study apparently free of other problems that could put it at a risk of bias?

Measures of treatment effect

For normally distributed outcomes, we calculated summary estimates of treatment effects using the inverse variance method. For dichotomous outcomes (mortality, kidney recovery and adverse events) results were expressed as risk ratio (RR) with 95% confidence intervals (CI). Where continuous scales of measurement were used to assess the effects of treatment (length of stay, cost) the mean difference (MD) was used or the standardised mean difference (SMD) if different scales were used.

Dealing with missing data

Any further information required from the original author was requested by written correspondence (e.g. emailing corresponding author) and any relevant information obtained in this manner was included in the review. Evaluation of important numerical data such as screened, randomised patients as well as intentionto-treat, as-treated and per-protocol population was carefully performed. Attrition rates, for example drop-outs, losses to followup and withdrawals were investigated. Issues of missing data and

imputation methods (e.g., last-observation-carried-forward) were critically appraised (Higgins 2011).

Assessment of heterogeneity

Heterogeneity was analysed using a Chi^2 test on N-1 degrees of freedom, with an alpha of 0.05 used for statistical significance and with the I² test (Higgins 2003). I² values of 25%, 50% and 75% correspond to low, medium and high levels of heterogeneity.

Assessment of reporting biases

If possible, funnel plots were to be used to assess the potential existence of small study bias (Higgins 2011).

Data synthesis

Data were pooled using the random-effects model but the fixedeffect model was also used to ensure robustness of the model chosen and susceptibility to outliers.

Subgroup analysis and investigation of heterogeneity

Subgroup analysis was used to explore possible sources of heterogeneity (such as intervention, participant and study quality). Heterogeneity among participants could relate to age, gender, fluid overload (< 10% and > 10% body weight relative to baseline), Intensive CRRT for AKI in homogenous subpopulations such as cardiac surgery or sepsis patients, effects of intensive continuous therapy on severity of illness - high, intermediate and low. We used appropriate scores of illness severity, such as Pediatric Risk of Mortality (PRISM), Pediatric Index of Mortality (PIM), Acute Physiology and Chronic Health Evaluation (Apache), Sequential Organ Failure Assessment (SOFA), and Cleveland Clinic ICU Acute Renal Failure (CCF).

Sensitivity analysis

We performed sensitivity analyses to explore the influence of the following factors on effect size.

- Repeating the analysis excluding unpublished studies
- Repeating the analysis taking account of risk of bias

• Repeating the analysis excluding any very long or large studies to establish how much they dominate the results

• Repeating the analysis excluding studies using the following filters: diagnostic criteria, language of publication, source of funding (industry versus other), and country.

'Summary of findings' tables

We presented the main results of the review in 'Summary of findings' tables. These tables present key information concerning the quality of the evidence, the magnitude of the effects of the interventions examined, and the sum of the available data for the main outcomes (Schünemann 2011a). The 'Summary of findings' tables also include an overall grading of the evidence related to each of the main outcomes using the GRADE (Grades of Recommendation, Assessment, Development and Evaluation) approach (GRADE 2008). The GRADE approach defines the quality of a body of evidence as the extent to which one can be confident that an estimate of effect or association is close to the true quantity of specific interest. The quality of a body of evidence involves consideration of within-study risk of bias (methodological quality), directness of evidence, heterogeneity, precision of effect estimates and risk of publication bias (Schünemann 2011b). Two summary of findings tables were created. Summary of findings for the main comparison summarizes the main findings for the comparison "Intensive versus less intensive RRT for acute kidney injury". We presented the following outcomes.

- Mortality until day 30 post-randomisation
- Mortality after 30 days post-randomisation

• Kidney function recovery: number of patients free of RRT after discontinuing CRRT

 Kidney function recovery: number of patients free of RRT after discontinuing CRRT until day 90, among survivals

• Adverse events: number of patients with

hypophosphataemia

Summary of findings 2 summarizes the main mortality findings for the subgroups of patients with AKI with and without sepsis, and related or not to surgery.

RESULTS

Description of studies

See. Characteristics of included studies; Characteristics of excluded studies; Characteristics of ongoing studies.

Results of the search

We identified 185 records from electronic databases (MEDLINE, EMBASE, CENTRAL, Cochrane Kidney and Transplant's Specialised Register, LILACS) to 9 February 2016. We screened these titles and abstracts excluded 117 records. We assessed the full text of 68 potentially eligible records (21 studies). Six studies (52 records) were included in our review (ATN Study 2005; Bouman 2002; RENAL Study 2006; Ronco 2000a; Saudan 2006; Tolwani 2008). One study has recently been completed but no results have been published (NCT01560650). Fourteen studies (15 records) were excluded (Figure 1). There was no disagreement among authors regarding inclusion of studies.



Figure I. Study flow diagram

Included studies

Six included studies (ATN Study 2005; Bouman 2002; RENAL Study 2006; Ronco 2000a; Saudan 2006; Tolwani 2008) enrolled a total of 3185 participants.

Study participants were all admitted to ICU. The mean age ranged from 51 and 68 years, and the proportion of male ranged from 55% to 71%. Sepsis was the primary cause of AKI in four studies (ATN Study 2005; RENAL Study 2006; Saudan 2006; Tolwani 2008) and surgery or cardio-surgery the main cause in the other two (Bouman 2002; Ronco 2000a)

All studies were reported between 2000 and 2008. Three were single-centre studies (Ronco 2000a; Saudan 2006; Tolwani 2008) and three were multicentre (ATN Study 2005; Bouman 2002; RENAL Study 2006).

Four studies used one CRRT modality exclusively; RENAL Study 2006 and Tolwani 2008 used CVVHDF, and Bouman 2002 and Ronco 2000a used CVVH. Saudan 2006 used CVVH and CVVHDF and ATN Study 2005 used intermittent haemodialysis (IHD) and CVVHDF or sustained low-efficiency haemodialysis (SLED), depending on the haemodynamic stability of the participant. Replacement fluid was administered either pre filter (ATN Study 2005; Saudan 2006; Tolwani 2008) or post filter (Bouman

2002; Ronco 2000a; RENAL Study 2006) when either CVVH or CVVHDF were used.

Five studies assessed the effects of two intensities of continuous therapy (ATN Study 2005; Bouman 2002; RENAL Study 2006; Saudan 2006; Tolwani 2008), whereas one assessed the effects of three CRRT intensities (standard, intermediate and high) (Ronco 2000a). For the purpose of the analysis, we combined the intermediate with high-dose arm of this study to create one high intensity arm. In Bouman 2002, two arms received the same less intensive CRRT dose but differed only in the timing of CRRT initiation. We combined these two treatment arms to create one less intensive arm. ATN Study 2005 randomly assigned critically ill patients with AKI to high-intensity or low-intensity RRT. Within treatment groups, patients were allocated to intermittent (IHD) or prolonged (SLED) and continuous RRT according to cardiovascular SOFA score. Continuous RRT was provided to 69.7% of patients as their initial therapy. However, unstable patients assigned to CRRT had a variable number of switches in treatment modality: none (36.1%) or 1 (24%) and \geq 2 modalities (10%) (Palevsky 2009). For the purpose of the analysis, we included all patients initially allocated to continuous CRRT independently of the switches in treatment modality (intention-to-treat analysis), who survived to day 60.

Five studies prescribed dose according to patients' weight at the time of admission (ATN Study 2005; RENAL Study 2006; Ronco 2000a; Saudan 2006; Tolwani 2008), and only one prescribed dose per unit of time (Bouman 2002).

Overall, the prescribed dose of CRRT in the less intensive arm of included studies ranged between 20 to 25 mL/kg/h and in intensive arm ranged between 35 to 48 mL/kg/h.

For full details see Characteristics of included studies.

Excluded studies

Fourteen studies were excluded. Cole 2002 and Vesconi 2009 compared continuous dialysis therapy versus no haemofiltration or other RRT (intermittent haemodialysis). Boussekey 2008,

IVOIRE Study 2013, Sanchez 2010b and Zhang 2012 and two ongoing studies (NCT01191905; NCT01251081) compared different intensity-arms treatment. Ghani 2006 and Payen 2009 did not provide relevant outcomes for this review. Two studies were not RCTs (Brause 2003; Zha 2012). In HEROICS Study 2015, 36% of control-arm patients did not receive CRRT. Jiang 2005 had different inclusion criteria in relation to our review. Only six patients (16%) with severe pancreatitis have AKI and were treated with CRRT (Characteristics of excluded studies).

Risk of bias in included studies

Included studies were generally at low or unclear risk of bias for all domains (See Figure 2; Figure 3).

Figure 2. Risk of bias graph: review authors' judgements about each risk of bias item presented as percentages across all included studies



| | Random sequence generation (selection bias) | Allocation concealment (selection bias) | Blinding of participants and personnel (performance bias) | Blinding of outcome assessment (detection bias) | Incomplete outcome data (attrition bias) | Selective reporting (reporting bias) | Other bias |
|------------------|---------------------------------------------|-----------------------------------------|-----------------------------------------------------------|-------------------------------------------------|------------------------------------------|--------------------------------------|------------|
| ATN Study 2005 | • | • | ? | • | • | Ŧ | • |
| Bouman 2002 | ÷ | • | ? | ÷ | ŧ | ÷ | ? |
| RENAL Study 2006 | • | • | ? | • | • | • | • |
| Ronco 2000a | • | • | ? | • | • | + | ? |
| Saudan 2006 | • | ? | ? | • | • | • | ? |
| Tolwani 2008 | • | • | ? | • | • | • | • |

Figure 3. Risk of bias summary: review authors' judgements about each risk of bias item for each included study

Allocation

Bouman 2002 and Ronco 2000a did not provide detailed information about random sequence generation and allocation concealment processes. Authors were contacted and we were informed that the random sequence generation was appropriate (computergenerated) and sealed opaque envelopes were used for the allocation process. All studies were assessed as being at low risk of selection bias due to appropriate random sequence generation and five were allocation concealment processes. The allocation process was considered unclear in Saudan 2006 due to disease severity imbalance observed between the study arms.

Blinding

All included studies were assessed at low risk of detection bias (outcome measurement was unlikely to be influenced by lack of blinding), and unclear risk of performance bias (insufficient information to enable judgment).

Incomplete outcome data

The risk of attrition bias was low in all included studies. Intentionto-treat analysis was performed in all studies.

Selective reporting

Published records included all expected outcomes and were considered at low risk of bias.

Other potential sources of bias

Three studies provided complete information on grant support for authors and studies (ATN Study 2005; RENAL Study 2006; Tolwani 2008). This information was unclear in the remaining three studies.

Effects of interventions

See: Summary of findings for the main comparison Intensive versus less intensive continuous renal replacement therapy (CRRT) for acute kidney injury (AKI); Summary of findings 2 Intensive versus less intensive continuous renal replacement therapy (CRRT) for acute kidney injury (AKI): subgroups The effects of intensive CRRT versus less intensive CRRT for main results and the quality of the evidence are summarised in Summary of findings for the main comparison

Mortality

All six studies assessed the effect of different intensities of CRRT (intensive versus less intensive treatment) on mortality. These studies varied in mortality reporting timing: at 90 days (RENAL Study 2006; Saudan 2006), 60 days (ATN Study 2005), 28 days after randomisation or at ICU discharge (Bouman 2002; Tolwani 2008), and 15 days after cessation of CRRT (Ronco 2000a). There was no significant difference between intensive versus less

intensive CRRT on mortality risk at day 30 (Analysis 1.1.1 (5 studies, 2402 participants): RR 0.88, 95% CI 0.71 to 1.08; $I^2 = 75\%$) or after 30 days post randomisation (Analysis 1.1.2 (5 studies, 2759 participants): RR 0.92, 95% CI 0.80 to 1.06; $I^2 = 65\%$). There was substantial heterogeneity among studies. We downgraded the quality of evidence from high to low due to this inconsistency and imprecision (Summary of findings for the main comparison).

Subgroup analysis and investigation of heterogeneity

There was evidence of significant heterogeneity in the magnitude of the effect among the included studies that measured mortality at different times of randomisation (at day 30 and after 30 days). To assess heterogeneity among participants we planned to perform the following pre-specified subgroup analyses: by age, gender, fluid overload ($\leq 10\%$ and > 10% in body weight relative to baseline), according to aetiology of AKI and severity of illness. Only data for aetiology of AKI and severity of illness were available.

The effect of the severity of illness at baseline was assessed using two subgroups: patients with and without sepsis and patients with high and low SOFA cardiovascular scores (\geq 3 and < 3). The was no significant differences in mortality between intensive versus less intensive CRRT in patients with sepsis (Analysis 1.2.1 (5 studies, 966 participants): RR 0.94; 95% CI 0.69 to 1.27; I² = 72%) or without sepsis (Analysis 1.2.2 (4 studies, 1216 participants): RR 0.89, 95% CI 0.69 to 1.15; I² = 73%), or in patients with SOFA scores < 3 (Analysis 1.2.3; 1 study, 404 participants: RR 0.91, 95% CI 0.71 to 1.18) or SOFA score \geq 3 (Analysis 1.2.4; 1 study, 1056 participants: RR 1.04, 95% CI 0.92 to 1.18).

The effect of AKI aetiology was considered using two subgroups: patients with AKI secondary to surgical causes and patients with AKI related to non-surgical causes. Compared to less intensive CRRT, Intensive CRRT reduced the risk of death in patients with post-surgical AKI (Analysis 1.2.5 (2 studies, 531 participants): RR 0.73, 95% CI 0.61 to 0.88; $I^2 = 0\%$), but not in patients with AKI related to non-surgical causes (Analysis 1.2.6 (3 studies, 1871 participants): RR 0.94, 95% CI 0.73 to 1.20; $I^2 = 76\%$).

The heterogeneity observed in subgroup analyses could be explained by AKI actiology (test for subgroup differences: $\text{Chi}^2 = 9.56$; P = 0.09; I² = 47.7%). When the post-surgery AKI group

was removed from the analysis, $I^2 = 0\%$ (test for subgroup differences: $Chi^2 = 1.96$, P = 0.74).

We downgraded the quality of evidence from high to low due to inconsistency and imprecision on the following subgroup analyses: patients with and without sepsis, patients with high and low SOFA cardiovascular score and patients with non-surgical AKI. High quality of evidence was found for the subgroup of patients with AKI related to surgery (Summary of findings 2).

Sensitivity analysis

The sensitivity analysis was performed excluding studies by risk of bias and large studies. When the analysis was developed taking risk of bias into account we observed that Saudan 2006 contributed to heterogeneity, and when excluded, heterogeneity was not significant (P = 0.63; $I^2 = 0\%$). The reason for exclusion was significant imbalance in the severity of illness observed between treatment arms. The effect of high intensity on mortality changed, but the direction of effects remained constant. We found no changes in heterogeneity when the study with larger sample size was excluded. Data on death or non-recovery at 90 days was not available.

Recovery of kidney function

Five studies reported information on recovery of kidney function (in all patients and among survivors). Studies varied in reporting of kidney recovery timing: 90 days after randomisation (RENAL Study 2006; Saudan 2006); 28 days (RENAL Study 2006; Saudan 2006; Tolwani 2008); at hospital discharge (Bouman 2002; Tolwani 2008); or 15 days after cessation of CRRT (Ronco 2000a).

Overall, there was no significant difference between intensive versus less intensive CRRT in the numbers of patients who were free of RRT after CRRT discontinuation (Analysis 1.3.1 (5 studies, 2402 participants): RR 1.12, 95% CI 0.91 to 1.37; $I^2 = 71\%$). There was substantial heterogeneity among studies. We downgraded the quality of evidence from high to low due to inconsistence and imprecision.

Similarly there was no significant difference between intensive versus less intensive CRRT on recovery of kidney function among survivors who discontinued CRRT at day 30 (Analysis 1.3.2 (5 studies, 1415 participants): RR 1.03, 95% CI 0.96 to 1.11; $I^2 =$ 69%) or at day 90 (Analysis 1.3.3 (3 studies, 988 participants): RR 0.98, IC 95% 0.94 to 1.01, $I^2 = 0$ %). We downgraded the quality of evidence from high to low due to inconsistency and indirectness and rated as moderate quality of evidence, due to indirectness, respectively (Summary of findings for the main comparison)

Subgroup analysis and investigation of heterogeneity

There was evidence of significant heterogeneity in the magnitude of the effect among the included studies that measured recovery of kidney function at different times in relation to randomisation. To assess heterogeneity among participants, we performed prespecified subgroup analyses by age, gender, fluid overload ($\leq 10\%$ and > 10% in body weight in relation to baseline), according to AKI aetiology and severity of illness. Only data for AKI aetiology were available. The effect of AKI aetiology was assessed using subgroups: patients with AKI predominantly related to surgical causes and patients with AKI related to non-surgical causes. Compared to less intensive CRRT, intensive CRRT increased recovery of kidney function in patients with post-surgical AKI (Analysis 1.4.1 (2 studies, 531 participants): RR 1.27, 95% CI 1.05 to 1.53, P = 0.01, $I^2 = 0\%$), but there were no difference on recovery of kidney function in patients with AKI related to non-surgical causes (Analysis 1.4.2 (3 studies, 1870 participants): RR 1.12, 95% CI 0.73 to 1.71, P = 0.61, I^2 = 82%). There was no heterogeneity between groups (test for subgroup differences: $Chi^2 = 0.27$, P = $0.60, I^2 = 0\%$).

Sensitivity analysis

There was evidence of heterogeneity in recovery of kidney function up to 30 days after discontinuation of CRRT ($I^2 = 69\%$). We performed sensitivity analyses to explore the above listed factors on the effect size. Few data were available. Sensitivity analysis was performed excluding studies with high risk of bias and those with large sample sizes. When the analysis was developed taking risk of bias into account, we observed that Ronco 2000a contributed to heterogeneity. When excluded, heterogeneity was not significant (P = 0.63; $I^2 = 0\%$). Ronco 2000a included a non-validated outcome (kidney recovery 15 days after cessation of CRRT) and enrolled patients with high incidence of post-surgical AKI, which may have contributed to a better prognosis on recovery of kidney function. The effect of intensive CRRT remained constant and the direction of effects did not change.

Length of stay

RENAL Study 2006 and Tolwani 2008 compared the effects of intensity of CRRT on length of stay. There were no significant differences between intensive and less intensive CRRT on the number of days in hospital (Analysis 1.5.1 (2 studies, 1665 participants): MD -0.23 days, 95% CI -3.35 to 2.89; $I^2 = 8\%$) and the number of days in ICU (Analysis 1.5.2 (2 studies, 1665 participants): MD -0.58 days, 95% CI -3.73 to 2.56, $I^2 = 19\%$). We downgraded the quality of evidence from high to low due to substantial imprecision and risk of bias.

Metabolic control

There was no significant difference between intensive and less intensive CRRT on the numbers of patients who normalised metabolic acidosis (Analysis 1.6.1 (1 study, 115 participants): RR 1.05, 95% CI 0.73 to 1.51). We rated this as moderate quality evidence due to imprecision.

Adverse events

The effects of intensity of CRRT on adverse events were reported in three studies (Bouman 2002; RENAL Study 2006; Saudan 2006). Intensive CRRT increased the risk of hypophosphataemia (Analysis 1.7.2 (1 study, 1441 participants): RR 1.21, 95% CI 1.11 to 1.31) compared to less intensive CRRT. We rated this as high quality evidence.

There were no significant differences between intensive and less intensive CRRT on the numbers of patients who experienced adverse events (Analysis 1.7.1 (3 studies, 1753 participants): RR 1.08, 95% CI 0.73 to 1.61; $I^2 = 16\%$), hypokalaemia (Analysis 1.7.3 (1 study, 1455 participants): RR 0.96, 95% CI 0.80 to 1.15), arrhythmia (Analysis 1.7.4 (1 study, 1463 participants): RR 0.92, 95% CI 0.80 to 1.06), and bleeding (Analysis 1.7.5 (3 studies, 1.7.5 (3 stu

1775 participants): RR 0.78, 95% CI 0.27 to 2.24; $I^2 = 0\%$). We downgraded the quality of evidence from high to moderate due to imprecision and for the last outcome we rate as low quality evidence due to substantial imprecision (Summary of findings for the main comparison).

Fluid balance and costs of CRRT were not reported in any of the included studies.

Evaluation of publication bias

We constructed a funnel plot to investigate potential publication bias. Meta-analysis of mortality at day 30 was analysed. We found reasonable symmetry indicating a low risk of publication bias (Figure 4).





Intensive versus less intensive CRRT for AKI: subgroups Patient or population: patients with AKI who need CRRT Settings: ICU Intervention: Intensive CRRT Comparison: Less intensive CRRT Illustrative comparative risks* (95% CI) No. of participants Quality of the evidence Comments **Relative effect** Outcomes (95% CI) (studies) (GRADE) Corresponding risk Assumed risk Standard dose High dose Mortality: patients with Study population RR 0.94 966 (5) $\oplus \oplus \bigcirc \bigcirc$ sepsis (0.69 to 1.27) low^{1,2} Follow-up: mean 28 524 per 1000 492 per 1000 days (361 to 665) Moderate 618 per 1000 581 per 1000 (426 to 785) Mortality: patients Study population RR 0.89 1216 (4) $\oplus \oplus \bigcirc \bigcirc$ without sepsis (0.69 to 1.15) **low**^{1,2} 414 per 1000 Follow-up: mean 28 465 per 1000 (321 to 535) days Moderate 564 per 1000 502 per 1000 (389 to 649)

ADDITIONAL SUMMARY OF FINDINGS [Explanation]

Intensity of continuous renal replacement therapy for acute kidney injury (Review) Copyright © 2016 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

9

| Mortality: patients with | Study population | | RR 0.73 | 531 (2) | ⊕⊕⊕⊕ biab |
|-------------------------------------------------------------|------------------|-------------------------------------|----------------|----------|-------------------------------|
| or general surgery Follow-up: mean 21 | 505 per 1000 | 368 per 1000 (308 to 444) | (0.6110 0.88) | | nign |
| uays | Moderate | | | | |
| | 459 per 1000 | 335 per 1000 (280 to 404) | | | |
| Mortality: patients with | Study population | | RR 0.94 | 1871 (3) | $\Phi \Phi \bigcirc \bigcirc$ |
| AKI not related to surgery Follow-up: mean 30 down | 414 per 1000 | 389 per 1000 (302 to 497) | (0.73 to 1.20) | | low ^{1,2} |
| uays | Moderate | | | | |
| | 550 per 1000 | 517 per 1000 (402 to 660) | | | |

*The basis for the **assumed risk** (e.g. the median control group risk across studies) is provided in footnotes. The **corresponding risk** (and its 95% confidence interval) is based on the assumed risk in the comparison group and the **relative effect** of the intervention (and its 95% Cl). **Cl:** Confidence interval/s; **RR:** Risk ratio

GRADE Working Group grades of evidence

High quality: Further research is very unlikely to change our confidence in the estimate of effect.

Moderate quality: Further research is likely to have an important impact on our confidence in the estimate of effect and may change the estimate.

Low quality: Further research is very likely to have an important impact on our confidence in the estimate of effect and is likely to change the estimate. Very low quality: We are very uncertain about the estimate.

¹ inconsistency: due to substantial heterogeneity (l² values ranged from 73% to 78%) ² imprecision: due to wide CI which crossed the threshold for clinically meaningful effects

DISCUSSION

Summary of main results

Our systematic review and subsequent meta-analysis examines the effect of different intensities of CRRT on mortality, kidney recovery function and adverse events among 3185 critically ill patients with AKI. Most of the included studies were assessed at low or unclear risk of bias for all domains. Within the intensity ranges assessed, more intensive CRRT did not demonstrate beneficial effects on mortality at different time points post randomisation (day 30 or after 30 days) and kidney recovery function (in all patients and among survivals at 30 and 90 days before randomisation) compared with less intensive therapy. The overall estimated effects on these outcomes are not statistically significant; the confidence intervals are sufficiently wide to include clinically benefits and harm (imprecision), with high level of heterogeneity (inconsistency). Treatment with more intensive CRRT increased the risk of hypophosphataemia, although it did not increase other adverse events when compared to less intensive CRRT. All results (except hypophosphataemia) were imprecise because the confidence intervals were wide which crossed the threshold for clinically meaningful effects.

An important limitation of this systematic review was the substantial heterogeneity found in the main results as mortality at day 30 or after 30 days ($I^2 = 75\%$ and $I^2 = 65\%$ respectively) and recovery of kidney function in all patients and among survivals at 30 ($I^2 = 71\%$ and $I^2 = 69\%$ respectively). However, there was no heterogeneity identified for recovery of kidney function at 90 days ($I^2 = 0\%$). We explored this heterogeneity by two prespecified clinical subgroup analyses; severity of acute illness (patients with and without sepsis and high and low SOFA cardiovascular scores) and by aetiology of AKI. We found that in patients with surgery-acquired AKI, intensive CRRT reduced mortality risk at day 30 compared to those patients with non-surgically-acquired AKI. Therefore, AKI aetiology was identified as a source of heterogeneity in the size of effect among included studies.

More intensive CRRT had uncertain effects on length of stay, number of days in ICU and number of days in hospital. These results should be interpreted with caution owing to the fact that only two small studies reported these data. Some studies have reported days in hospital and days in ICU but, in patients with a high short-term mortality risk, the interpretation of such results may be misleading given the mortality is a competing end point for length of stay.

Overall completeness and applicability of evidence

See Description of studies; Characteristics of included studies Six randomised studies (ATN Study 2005; Bouman 2002; RENAL Study 2006; Ronco 2000a; Saudan 2006; Tolwani 2008) evaluated the effect of different intensities of CRRT on survival and recovery of kidney function in critically ill patients with AKI. Two studies (Ronco 2000a; Saudan 2006) favoured more intensive therapy. In contrast, four other studies (ATN Study 2005; Bouman 2002; RENAL Study 2006; Tolwani 2008) have not demonstrated beneficial effects with an increased intensity of therapy on clinical outcomes. These results are consistent with those reported in our review.

Disparity and the heterogeneity found in these results among the studies probably may explained by several factors such as differences in methodological quality of studies, patients characteristics, delivered dialysis dose, and timing of CRRT initiation.

In a study of 425 patients, Ronco 2000a reported a decrease in mortality at day 15 from 59% to 43% with an increased in the intensity from 20 mL/kg/h to 35 or 45 mL/kg/h. We observed some limitations in this study: an absence of detailed description of randomisation and allocation concealment process (limiting the internal validity); a low incidence of patients with AKI-related to sepsis (15%); and a non-validated short-term outcome (limiting the external validity). In Saudan 2006 (206 patients with AKI) there was a 26% reduction in all-cause of mortality at day 90 (from 62% to 36%) with an increase in the intensity of CRRT from 25 mL/kg/h to 45 mL/kg/h. This study has an important imbalance in the severity of illness observed between CVVH treatment arms (limiting the internal validity). Additionally, both studies were unblinded, single-centre studies (limiting the internal and external validity respectively). In contrast, three studies evaluating intensity in continuous therapy, and one study in combined modalities (Intermittent and continuous), did not demonstrate any effect of increased intensity of therapy on survival. Bouman 2002 conducted a small study evaluating both intensity and timing of initiation of CVVH in 106 critical patients with AKI. There were no differences on the survival for either intensities or initiation time. It is interesting to note that the actual delivered therapy in the high-intensity arm was much less than the prescribed intensity. Furthermore, survival was greater than expected (survival at 28 days 69% to 75% in all groups) probably related to a low incidence of patients with AKI-related to sepsis (limiting the external validity). The study was underpowered due to the small sample size. Similarly, Tolwani 2008 evaluated 200 patients with AKI. They found no difference on survival with intensive continuous therapy. This was an unblinded, single-centre study (limiting the internal and external validity respectively). Finally, two large multicentre studies were conducted. In ATN Study 2005, 1124 critically ill patients with AKI were randomised to high-intensity or low-intensity. Within treatment groups, patients were allocated to CVVHDF or SLED and IHD according to cardiovascular SOFA score. In RENAL Study 2006, 1508 patients were randomly assigned to two intensities of CVVHDF (intensive or less intensive). All patients received CRRT as their first mode of RRT; only a small proportion of patients received IHD (7%). Both studies reported no beneficial effect on mortality and recovery of kidney function associated with a more intensive RRT.

There were also differences in the prescribed and delivered doses

of CRRT. In ATN Study 2005, the dose delivered was 89% of that prescribed for higher-intensity treatment, Tolwani 2008 reported a value of 83%, and in RENAL Study 2006 the delivered dose was 84%. For the lower-intensity treatment, the doses delivered were 95% in ATN Study 2005, 85% in Tolwani 2008, and 88% in RENAL Study 2006. In all other studies (Bouman 2002; Ronco 2000a; Saudan 2006) delivered doses were less than 85% of the prescribed doses. The difference between the prescribed and the delivered dose highlights the risk of overestimating the effective delivery of therapy and the need to improve operational measures in CRRT.

Although the analysis included data obtained from a comprehensive and rigorous search, we identified gaps in several areas. The majority of participants of the included studies were adults, limiting the applicability of our finding to children. In general, the incidence of AKI secondary to sepsis is very high in ICU (50% to 60%); however, in two studies it was observed that the majority of patients had post-surgical AKI, and relatively few had sepsis or preexisting chronic kidney disease, limiting the applicability of our results to general ICU population. Three included studies were single-centre studies, limiting the external validity of the results. While the urea kinetics remains widely used to measure intensity of RRT in AKI-patients, this approach provides an incomplete assessment of dose of RRT, especially in the critically ill patients with AKI.

An important challenge when examining the evidence of dialysis intensity in patients with AKI is to determine the exact number of patients who received IHD or CRRT in those studies using a combination of both strategies (ATN Study 2005), as well as distinguishing which patients remained dialysis-depend after ICU discharge or received transitory IHD without regard to the original assigned treatment. In view of this, we contacted the authors for more information. The long-term kidney outcomes after hospital discharge among survivors of AKI remain poorly characterised. The studies did not report data on mortality and kidney function recovery in patients with pre-existing chronic kidney disease and with low or intermediate scores of severity of illness. The results on length of stay (days in hospital and in the ICU) and recovery of kidney function should be interpreted with caution, especially when the mortality risk is taken into account.

We are aware that an important aspect to consider in term of efficacy is the timing in which CRRT are indicated. Currently, we are trying to answer this question with a systematic review specified (Fayad 2013a).

We included only RCTs with the purpose of reducing bias.

Quality of the evidence

We conducted this review according to the process described in the *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011). Our review was based on evidence from six RCTs (3185 participants) that compared different intensities of CRRT in critically ill patients with AKI. The quality of evidence for our main outcomes was drawn from studies assessed at low risk of bias for random sequence generation and allocation concealment processes, incomplete outcome data, intention to treat analysis, selective outcomes reporting, performance bias and other sources of bias; and unclear risk for detection bias.

Data comparing the effect of higher intensity versus lower intensity CRRT on mortality at day 30 or after were obtained from six well-conducted RCTs respectively, but we downgraded the quality of evidence to low, mainly due to inconsistencies (I² values of 65% and 75%) and imprecision (CIs included a range of plausible values with clinically important benefits but also included harms). Similarly, we downgraded the quality of evidence to low for recovery of kidney function in all patients and among survivors at day 30 due to inconsistencies (I² values of 71% and 69%) and rated as moderate data obtained for recovery of kidney function among survivors at day 90 by indirectness (the recovery of kidney function in this high risk group is affected when the risk of death is taken into account).

Data used to assess the impact of intensive versus less intensive CRRT on adverse events were obtained from three well-conducted RCTs, providing treatment effects with clinically important harms; however, we downgraded the quality of evidence to moderate due to imprecision (CIs included both clinically important benefits and harm). One study provided data on hypophosphataemia. We rated this as high quality evidence.

Potential biases in the review process

While this review was conducted according to rigorous methods developed by the Cochrane Collaboration, some bias may be present in the review process. We searched for all relevant studies using sensitive and validated strategies in major medical databases and grey literature sources. However, it is possible that some studies (such as unpublished data and studies with negative or no effects) were not identified. An analysis for evidence to assess the risk of publication bias was not possible for all outcomes due to the small number of studies available in each meta-analysis (Figure 4).

It was difficult to identify the number of patients who received IHD in the included study using a combination of both therapies (intermittent and continuous), as well also in studies evaluating CRRT intensity, in which patients remained dialysis-dependent after ICU or hospital discharge, many were likely to have transitioned to IHD regardless of the original study-assigned dose of CRRT.

Several subgroup analyses were planned to explore potential sources of heterogeneity in our review, however a lack of data prevented us from doing these analyses.

Agreements and disagreements with other studies or reviews

Our systematic review in keeping with previous meta-analysis on intensity in CRRT (Negash 2011) or in mixed modality, combining intermittent, sustained and continuous dialysis (Jun 2010; Van Wert 2010), has not found beneficial effects of more intensive RRT with respect to mortality and kidney recovery function in critically ill patients with AKI compared to less intensive therapy. There has been increased interest in recovery of kidney function. Indeed, lack of recovery of kidney function implies the need for long-term dialysis associated with low quality of life. Our review has not demonstrated benefits on recovery of kidney function with intensive therapy. These findings are consistent with four individual RCTs (ATN Study 2005; Bouman 2002; RENAL Study 2006; Tolwani 2008) and do not agree with those reported of two previous RCTs (Ronco 2000a; Saudan 2006). It is important to note that relevant differences on recovery of kidney function between ATN Study 2005 and RENAL Study 2006 were observed (45.2% versus 13.3% of survivors depend on RRT at day 28 respectively). These differences may be due to several factors including different populations, prevalence of intermittent dialysis, pre-existing chronic kidney disease and timing of RRT initiation. A review by Palevsky 2005 on factors affecting kidney recovery following AKI did not recommend either intensities with regard to recovery of kidney function when the mortality risk is taken into account (given that mortality is a competing end point for recovery of kidney function).

The hypothesis that in critically ill patients, especially those with sepsis or systemic inflammatory responses, could benefit from an intensive CRRT was proposed by several researchers. It is interesting to note that we did not find benefit from higher intensity CRRT in this subgroup of patients in our review. These results were consistent with previous meta-analysis (Jun 2010; Van Wert 2010). Additionally, previous reviews explored the effect of high volume haemofiltration (HVHF) specifically in critically ill patients with severe sepsis or septic shock in an ICU setting (Borthwick 2013; Clark 2014; Lehner 2014). These reviews applied different thresholds for HVHF: Borthwick 2013 defined HVHF as > 35 mL/kg/h, while more recent reviews define HVHF as >50 mL/kg/h (Clark 2014) and HVHF and pulse high volume haemofiltration (PHVHF) as 85 mL/Kg/h (Lehner 2014). These reviews included studies we excluded from our review due to the very-high intensity applied (Boussekey 2008; IVOIRE Study 2013; Sanchez 2010b; Zhang 2012) or no requirement of AKI for enrolment (Jiang 2005) These reviews found insufficient evidence of a therapeutic benefit for routine use of HVHF for septic AKI.

AUTHORS' CONCLUSIONS

Implications for practice

With available data of included RCTs, more intensive CRRT (range between 35 to 48 mL/kg/h) demonstrated no beneficial

effects on mortality or recovery of kidney function, however there was an increased risk of hypophosphataemia compared to less intensive therapy (range between 20 to 25 mL/kg/h). The absence of high quality evidence of efficacy and the possibility of increased adverse events does not support the routine use of high intensity CRRT in this group of patients. However, in patients with postsurgical AKI, high intensity CRRT appears to reduce the risk of death.

These results do not minimise the importance of the intensity in continuous treatment of critically ill patients with AKI. Minimal standards for the delivered dialysis dose of therapy appear to have been identified (KDIGO 2012). There is evidence to suggest that the mortality in these high-risk populations will be substantially altered by improvements in the delivery kidney support.

Our results are likely to have implications for clinical practice in countries (Europe; New Zealand and Australia) where CRRT is now the preferred form of RRT in the ICU (Uchino 2005). However, in clinical practice, haemodynamically unstable patients are commonly managed using CRRT and haemodynamically stable patients are generally treated using IHD; frequently these patients receive both modalities over the course of their illness as their haemodynamic status change.

Implications for research

Given the persistently high mortality rate among patients with AKI, it would be important to accurately determine the effect of intensity of CRRT on mortality particularly in patients with postsurgical AKI. In view of the inconsistencies observed in the main outcomes and the inability to assess all possible causes of heterogeneity, it would be important to perform pooled analyses of individual patient data from all completed studies to deal with heterogeneity issues. Such an initiative (a patient-level meta-analysis of all of the intensity of RRT studies) is being conducted by the George Institute for Global Health in Australia.

Optimal timing of CRRT initiation during therapy needs to be rigorously evaluated. It would be important to perform pooled analysis of individual patient data from all completed studies to deal with heterogeneity issues.

There is also a need to investigate other strategies that can be implemented alone or concurrently to CRRT for the treatment of AKI.

ACKNOWLEDGEMENTS

The authors wish to thank the referees for their advice and feedback during the preparation of this review.

The authors would also like to thank all study authors who provided additional information about their studies and Daniel Co-

mande, Librarian of the Institute for Clinical Effectiveness and Health Policy (Argentina) for his help in search for the literature.

We would like to thank Ms Narelle Willis and Ms Gail Higgins from Cochrane Kidney and Transplant for their help with this review and especially to Marta Roque Figuls, Iberoamerican Cochrane Centre for her support and editorial advice during the preparation of this systematic review.

REFERENCES

References to studies included in this review

ATN Study 2005 {published data only}

Afshinnia F, Belanger K, Palevsky PM, Young EW. Effect of ionized serum calcium on outcomes in acute kidney injury needing renal replacement therapy: secondary analysis of the acute renal failure trial network study. *Renal Failure* 2013;**35**(10):1310–8. [MEDLINE: 23992422] Crowley S, Schein R, Dev D, Finkel K, Vijayan A, Paganini E, et al. Dialysis catheter complications in the VA/NIH ATN study [abstract no: SA-PO557]. *Journal of the American Society of Nephrology* 2007;**18**(Abstracts):463A. Crowley S, Schein R, Dev D, Finkel K, Vijayan A, Paganini E, et al. Lessons for successful study enrollment from the VA/NIH ATN study [abstract no: SA-PO932]. *Journal of the American Society of Nephrology* 2006;**17**(Abstracts): 770A. [CENTRAL: CN–00601950;]

Crowley ST, Chertow GM, Vitale J, O'Connor T, Zhang J, Schein RM, et al. Lessons for successful study enrollment from the Veterans Affairs/National Institutes of Health Acute Renal Failure Trial Network Study. *Clinical Journal of The American Society of Nephrology: CJASN* 2008;**3**(4): 955–61. [MEDLINE: 18385390]

Demirjian S, Paganini EP, Zhang JH, O'Connor TZ, Vitale J, Palevsky PM. Severity of illness does not modify the effect of intensity of renal replacement therapy (RRT) on outcome in critically ill patients with AKI: results from the VA/NIH acute renal failure trial network (ATN) study [abstract no: SA-PO2997]. *Journal of the American Society of Nephrology* 2008;**19**(Abstracts Issue):791A.

Demirjian S, Paginini EP, Zhang JH, O'Connor TZ, Vitale J, Palevsky PM, et al. Predictive scoring systems perform poorly in critically ill patients with AKI requiring renal replacement: data from the VA/NIH Acute Renal Failure Trial Network (ATN) study [abstract no: SA-PO3010]. *Journal of the American Society of Nephrology* 2008;**19** (Abstracts Issue):794A.

Johansen KL, Smith MW, Unruh ML, Siroka AM, O'Connor TZ, Palevsky PM, et al. Predictors of health utility among 60-day survivors of acute kidney injury in the Veterans Affairs/National Institutes of Health Acute Renal Failure Trial Network Study. *Clinical Journal of The American Society of Nephrology: CJASN* 2010 Aug;**5**(8): 1366–72. [MEDLINE: 20507953] Joyce VR, Smith MW, Johansen KL, Unruh ML, Siroka AM, O'Connor TZ, et al. Health-related quality of life as a predictor of mortality among survivors of AKI. *Clinical Journal of The American Society of Nephrology: CJASN* 2012; 7(7):1063–70. [MEDLINE: 22595826] Palevsky PM, Franchini R, O'Connor TZ, Zhang JH, VA/NIH Acute Renal Failure Trial Network. Recovery of kidney function in critically ill patients with acute kidney injury (AKI) treated with intensive versus less-

intensive renal replacement therapy (RRT) [abstract no: SA-PO2994]. *Journal of the American Society of Nephrology* 2008;**19**(Abstracts Issue):790A.

Palevsky PM, O'Connor T, Zhang JH, Star RA, Smith MW. Design of the VA/NIH Acute Renal Failure Trial Network (ATN) Study: intensive versus conventional renal support in acute renal failure. *Clinical Trials* 2005;**2**(5):423–35. [MEDLINE: 16317811]

Palevsky PM, O'Connor TZ, Chertow GM, Crowley ST, Zhang JH, Kellum JA, et al. Intensity of renal replacement therapy in acute kidney injury: perspective from within the Acute Renal Failure Trial Network Study. *Critical Care (London, England)* 2010;**13**(4):310. [MEDLINE: 19678919]

Palevsky PM, O'Connor TZ, Zhang JH, Star R. VA/NIH Acute Renal Failure Trial: study design [abstract no: SA-PO970]. *Journal of the American Society of Nephrology* 2003;**14**(Program & Abstracts):512A. [CENTRAL: CN–00583741;]

Palevsky PM, Overberger P, Franchini R, O'Connor TZ, Zhang JH, VA/NIH Acute Renal Failure Trial Network. One-year outcomes in critically ill patients with acute kidney injury (AKI) treated with intensive versus lessintensive renal replacement therapy (RRT) [abstract no: SA-FC414]. *Journal of the American Society of Nephrology* 2008;**19**(Abstracts Issue):93A.

Palevsky PM, Zhang J, O'Connor T. Intensive versus nonintensive renal replacement therapy (RRT) in critically ill patients with acute kidney injury (AKI) [abstract]. American Thoracic Society International Conference; 2008 May 16-21; Toronto, Canada. 2008:A767. [CENTRAL: CN–00716125;]

Pesacreta M, Overberger P, Palevsky PM, VA/NIH Acute Renal Failure Trial Network. Management of renal replacement therapy in acute renal failure: a survey of practitioner prescribing practices. [abstract no: SA-PO227].

Journal of the American Society of Nephrology 2004;**15**(Oct): 350A. [CENTRAL: CN–00601951;]

* VA/NIH Acute Renal Failure Trial Network, Palevsky PM, Zhang JH, O'Connor TZ, Chertow GM, Crowley ST, et al. Intensity of renal support in critically ill patients with acute kidney injury. [Erratum appears in N Engl J Med. 2009 Dec 10;361(24):2391]. *New England Journal of Medicine* 2008;**359**(1):7–20. [MEDLINE: 18492867] Zhang JH, O'Connor T, Palevsky PM, for the VA/NIH Acute Renal Failure Trial Network Study. Evaluation of treatment separation in the VA/NIH Acute Renal Failure Trial Network (ATN) Study [abstract]. *Clinical Trials (London, England)* 2009;**6**(5):523–4. [CENTRAL: CN–00783257;]

Zhang JH, O'Connor T, Swanson K, Palevsky PM, VA/ NIH Acute Renal Failure Trial Network Study. Evaluation of trial safety in an ICU trial: experience from the VA/ NIH Acute Renal Failure Trial Network (ATN) Study [abstract]. *Clinical Trials (London, England)* 2009;**6**(5):560. [CENTRAL: CN–00783258;]

Zhang JH, Palevsky PM, Chertow GM, Hartigan J, O'Connor TZ, Guarino P, et al. Piecewise analysis of patient survival after onset of AKI. *Clinical Journal of The American Society of Nephrology: CJASN* 2013;**8**(10): 1679–84. [MEDLINE: 23813558]

Bouman 2002 {published data only}

Bouman CS, Oudemans-Van Straaten H, Tijssen JG, Zandstra DF, Kesecioglu J. Effects of early high-volume continuous venovenous haemofiltration on survival and recovery of renal function in intensive care patients with acute renal failure: a prospective, randomized trial. *Critical Care Medicine* 2002;**30**(10):2205–11. [MEDLINE: 12394945]

RENAL Study 2006 {published data only}

Bellomo R. Do we know the optimal dose for renal replacement therapy in the intensive care unit?. Kidney International 2006;70(7):1202-4. [MEDLINE: 16988729] Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, Kim I, et al. The relationship between hypophosphataemia and outcomes during low-intensity and high-intensity continuous renal replacement therapy. [Erratum appears in Crit Care Resusc. 2014 Jun;16(2):139 Note: McGuiness, Shay [corrected to McGuinness, Shay]]. Critical Care & Resuscitation 2014;16(1):34-41. [MEDLINE: 24588434] Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, Lee J, et al. Calorie intake and patient outcomes in severe acute kidney injury: findings from The Randomized Evaluation of Normal vs. Augmented Level of Replacement Therapy (RENAL) study trial. Critical Care (London, England) 2014; 18(2):R45. [MEDLINE: 24629036]

Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, Lee J, et al. Daily protein intake and patient outcomes in severe acute kidney injury: findings of the randomized evaluation of normal versus augmented level of replacement therapy (RENAL) trial. *Blood Purification* 2014;**37**(4):325–34. [MEDLINE: 25171270] Bellomo R, Lipcsey M, Calzavacca P, Haase M, HaaseFielitz A, Licari E, et al. Early acid-base and blood pressure effects of continuous renal replacement therapy intensity in patients with metabolic acidosis. *Intensive Care Medicine* 2013;**39**(3):429–36. [MEDLINE: 23306586]

Finfer S, Cass A, Gallagher M, Lee J, Su S, Bellomo R, et al. The RENAL (Randomised Evaluation of Normal vs. Augmented Level of Replacement Therapy) study: statistical analysis plan. *Critical Care & Resuscitation* 2009; **11**(1):58–66. [MEDLINE: 19281446]

Gallagher M, Bellomo R, Cass A, Finfer S, Gattas D, Lee J, et al. Long term outcomes of severe AKI: results of the post-RENAL study [abstract no: 071]. *Nephrology* 2012;**17** (Suppl 2):45–6. [EMBASE: 71377309]

Gallagher M, Cass A, Bellomo R, Finfer S, Gattas D, Lee J, et al. Long-term survival and dialysis dependency following acute kidney injury in intensive care: extended follow-up of a randomized controlled trial. *PLoS Medicine* 2014;**11**(2): e1001601. [MEDLINE: 24523666]

Jun M, Bellomo R, Cass A, Gallagher M, Lo S. Timing of renal replacement therapy and patient outcomes in the randomized evaluation of normal vs augmented level of replacement therapy trial [abstract no: 008]. *Nephrology* 2012;**17**(Suppl 2):29–30. [EMBASE: 71377246] RENAL Replacement Therapy Study Investigators, Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, et al. An observational study fluid balance and patient outcomes in the Randomized Evaluation of Normal vs. Augmented Level of Replacement Therapy trial. *Critical Care Medicine* 2012;**40**(6):1753–60. [MEDLINE: 22610181] * RENAL Replacement Therapy Trial Investigators,

REINAL Replacement Therapy Trial Investigatols,
Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, et al. Intensity of continuous renal-replacement therapy in critically ill patients. *New England Journal of Medicine* 2009;
361(17):1627–38. [MEDLINE: 19846848]
RENAL Replacement Therapy Trial Investigators, Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, et al. Screening and study enrolment in the Randomized Evaluation of Normal vs. Augmented Level (RENAL) Replacement Therapy Trial. *Blood Purification* 2009;27(2):199–205. [MEDLINE: 19256108]

RENAL Study Investigators, Bellomo R, Cass A, Cole L, Finfer S, Gallagher M, et al. Design and challenges of the Randomized Evaluation of Normal versus Augmented Level Replacement Therapy (RENAL) Trial: high-dose versus standard-dose hemofiltration in acute renal failure. *Blood Purification* 2008;**26**(5):407–16. [MEDLINE: 18856012] Roberts D, Roberts M, Liu X, Roberts J, Lipman J, Bellomo R. Clearance of antibiotics by high and low intensity continuous renal replacement therapy in critically ill patients [abstract no: 232]. *Nephrology* 2010;**15**(Suppl 4): 87. [EMBASE: 70467236]

Roberts DM, Liu X, Roberts JA, Nair P, Cole L, Roberts MS, et al. A multicenter study on the effect of continuous hemodiafiltration intensity on antibiotic pharmacokinetics. *Critical Care (London, England)* 2015;**19** (1):84. [MEDLINE: 25881576]

Wang AY, Bellomo R, Ninomiya T, Lo S, Cass A, Jardine M,

et al. Angiotensin-converting enzyme inhibitor usage and acute kidney injury: a secondary analysis of RENAL study outcomes. *Nephrology* 2014;**19**(10):617–22. [MEDLINE: 24894685]

Ronco 2000a {published data only}

Ho TB, Jefferson HJ, Rhodes A. Continuous haemofiltration in acute renal failure. *Lancet* 2000;**356**(9239):1441–2. [MEDLINE: 11052612]

* Ronco C, Bellomo R, Homel P, Brendolan A, Dan M, Piccinni P, et al. Effects of different doses in continuous veno-venous haemofiltration on outcomes of acute renal failure: a prospective randomised trial. *Lancet* 2000;**356** (9223):26–30. [MEDLINE: 10892761]

Ronco C, Belomo R, Homel P, Brendolan A, Dan M, Piccinni P, et al. Effects of different doses in continuous veno-venous haemofiltration on outcomes of acute renal failure: a prospective randomised trial. *Edtna-Erca Journal* 2002;**Suppl 2**:7–12. [MEDLINE: 12371727]

Ronco C, Homel P, Bellomo R, Brendolan A. Prospective randomised trial on dose delivery versus outcomes of RF treated by continuous veno-venous hemofiltration (CVVH) [abstract no: A0717]. *Journal of the American Society of Nephrology* 2000;**11**(Sept):133A. [CENTRAL: CN–00644229;]

Schiffl H. Continuous haemofiltration in acute renal failure. *Lancet* 2000;**356**(9239):1441. [MEDLINE: 11052611] Than N, Turney JH. Continuous haemofiltration in acute renal failure. *Lancet* 2000;**356**(9239):1441. [MEDLINE: 11052610]

Saudan 2006 {published data only}

Saudan P, Niederberger M, De Seigneux S, Romand J, Pugin J, Perneger T, et al. A prospective randomized trial comparing continuous hemodiafiltration versus hemofiltration in critically ill patients with acute renal failure [abstract no: PUB003]. *Journal of the American Society of Nephrology* 2004;15(Oct):763A. [CENTRAL: CN-00724917;]

* Saudan P, Niederberger M, De Seigneux S, Romand J, Pugin J, Perneger T, et al. Adding a dialysis dose to continuous hemofiltration increases survival in patients with acute renal failure. *Kidney International* 2006;**70**(7): 1312–7. [MEDLINE: 16850022]

Saudan P, Niederberger M, Sellweger M, Pugin J, Romand J, Perneger T, et al. Continuous hemofiltration versus continuous hemodiafiltration in critically ill patients with acute renal failure [abstract no: PUB003]. *Nephrology Dialysis Transplantation* 2003;**18**(Suppl 4):666. [CENTRAL: CN–00447595;]

Saudan P, Triverio PA, Romand JA, Pugin J, Martin PY. Long-term prognosis in critically ill patients with acute renal failure treated by continuous renal replacement therapy [abstract no: TH-PO822]. *Journal of the American Society of Nephrology* 2006;**17**(Abstracts):282A. [CENTRAL: CN-00724918;]

Triverio PA, Martin PY, Romand J, Pugin J, Perneger T, Saudan P. Long-term prognosis after acute kidney injury requiring renal replacement therapy. *Nephrology*

Dialysis Transplantation 2009;**24**(7):2186–9. [MEDLINE: 19228754]

Tolwani 2008 {published data only}

Lyndon W, Wille K, Tolwani A. Solute clearance in CRRT: comparing measured effluent volume to actual delivered dose [abstract no: 177]. *American Journal of Kidney Diseases* 2011;**57**(4):A61. [CENTRAL: 70379736;]

Lyndon WD, Wille KM, Tolwani AJ. Solute clearance in CRRT: comparing measured effluent volume to actual delivered dose [abstract no: 26]. 16th International Conference on CRRT; 2011 Feb 22-25; San Diego, CA. 2011:127.

Lyndon WD, Wille KM, Tolwani AJ. Solute clearance in CRRT: prescribed dose versus actual delivered dose. *Nephrology Dialysis Transplantation* 2012;**27**(3):952–6. [MEDLINE: 21896498]

* Tolwani AJ, Campbell RC, Stofan BS, Lai KR, Oster RA, Wille KM. Standard versus high-dose CVVHDF for ICUrelated acute renal failure. *Journal of the American Society of Nephrology* 2008;**19**(6):1233–8. [MEDLINE: 18337480] Tolwani AJ, Speer R, Stofan B, Lai KR, Wille KM. A randomized prospective study comparing high dose continuous venovenous hemodiafiltration (CVVHDF) to standard CVVHDF in critically ill patients with acute renal injury [abstract no: 22]. *Blood Purification* 2007;**25**:193.

References to studies excluded from this review

Boussekey 2008 {published data only}

Boussekey N, Chiche A, Faure K, Devos P, Guery B, d'Escrivan T, et al. A pilot randomized study comparing high and low volume hemofiltration on vasopressor use in septic shock. *Intensive Care Medicine* 2008;**34**(9):1646–53. [MEDLINE: 18542921]

Brause 2003 {published data only}

Brause M, Neumann A, Schumacher T, Grabensee B, Heering P. Effect of filtration volume of continuous venovenous hemofiltration in the treatment of patients with acute renal failure in intensive care units. *Critical Care Medicine* 2003;**31**(3):841–6. [MEDLINE: 12626994]

Cole 2002 {published data only}

Cole L, Bellomo R, Hart G, Journois D, Davenport P, Tipping P, et al. A phase II randomized, controlled trial of continuous hemofiltration in sepsis. *Critical Care Medicine* 2002;**30**(1):100–6. [MEDLINE: 11902250]

Ghani 2006 {published data only}

Ghani RA, Zainudin S, Ctkong N, Rahman AF, Wafa SR, Mohamad M, et al. Serum IL-6 and IL-1-ra with sequential organ failure assessment scores in septic patients receiving high-volume haemofiltration and continuous venovenous haemofiltration. *Nephrology* 2006;**11**(5): 386–93. [MEDLINE: 17014550]

Zainudin S, Ghani RA, Mohd M, Wafa SR, Tong NK. Stability of haemodynamic parameters during CRRT: a comparison between standard continuous venovenous haemofiltration and high-volume haemofiltration in patients with acute renal failure and sepsis [abstract no:

SP286]. Nephrology Dialysis Transplantation 2006;21(Suppl 4):iv108. [CENTRAL: CN-00583762;]

HEROICS Study 2015 {published data only}

Combes A, Brechot N, Amour J, Cozic N, Lebreton G, Guidon C, et al. Early High-Volume Hemofiltration versus Standard Care for Post-Cardiac Surgery Shock. The HEROICS Study. *American Journal of Respiratory & Critical Care Medicine* 2015;**192**(10):1179–90. [MEDLINE: 26167637]

IVOIRE Study 2013 {published data only}

Joannes-Boyau O, Honore PM, Perez P, Bagshaw SM, Grand H, Canivet JL, et al. High-volume versus standardvolume haemofiltration for septic shock patients with acute kidney injury (IVOIRE study): a multicentre randomized controlled trial. *Intensive Care Medicine* 2013;**39**(9): 1535–46. [MEDLINE: 23740278]

Jiang 2005 {published data only}

Jiang HL, Xue WJ, Li DQ, Yin AP, Xin X, Li CH, et al. Influence of continuous veno-venous hemofiltration on the course of acute pancreatitis. *World Journal of Gastroenterology* 2005;**11**(31):4815–21. [MEDLINE: 16097050]

NCT01191905 {published data only}

Kim DK, Yoo TH. Effects of high volume continuous renal replacement therapy in patients with septic acute kidney injury. www.clinicaltrials.gov/ct2/show/NCT01191905 (accessed 25 August 2016).

NCT01251081 {published data only}

Chen J. Effect of the intensity of continuous renal replacement therapy in patients with sepsis and acute kidney injury: single-center randomized clinical trial. www.clinicaltrials.gov/ct2/show/NCT01251081 (accessed 25 August 2016).

Payen 2009 {published data only}

Payen D, Mateo J, Cavaillon JM, Fraisse F, Floriot C, Vicaut E, et al. Impact of continuous venovenous hemofiltration on organ failure during the early phase of severe sepsis: a randomized controlled trial. *Critical Care Medicine* 2009; **37**(3):803–10. [MEDLINE: 19237881]

Sanchez 2010b {published data only}

Sanchez C, Corbalan P, Rodriguez F, Sanchez A, Palominos S. Intensive high volume hemofiltration vs very high volume hemofiltration: effects on hemodynamics in patients with severe sepsis: a nursing approach [abstract no: 0432]. *Intensive Care Medicine* 2010;**36**(2 Suppl 2):S193.

Vesconi 2009 {published data only}

Vesconi S, Cruz DN, Fumagalli R, Kindgen-Milles D, Monti G, Marinho A, et al. Delivered dose of renal replacement therapy and mortality in critically ill patients with acute kidney injury. *Critical Care (London, England)* 2009;**13**(2):1–14. [MEDLINE: 19368724]

Zha 2012 {published data only}

Zha Y, Yang X, Lin X, Yuan J, Hu Y, Long YJ, et al. Clinical observation of different doses of continuous renal replacement therapy for severe pneumonia with acute kidney injury. *Chung-Hua i Hsueh Tsa Chih [Chinese Medical Journal]* 2012;**92**(48):3385–8. [MEDLINE: 23327695]

Zhang 2012 {published data only}

Zhang P, Yang Y, Lv R, Zhang Y, Xie W, Chen J. Effect of the intensity of continuous renal replacement therapy in patients with sepsis and acute kidney injury: a singlecentre randomized clinical trial. *Nephrology Dialysis Transplantation* 2012;**27**(3):967–73. [MEDLINE: 21891773]

References to studies awaiting assessment

NCT01560650 {published data only}

Shi W. Effect of the intensity of continuous renal replacement therapy in patients with acute kidney injury: single-centre randomised clinical trial. www.clinicaltrials.gov/ct2/show/ NCT01560650 (accessed 25 August 2016).

Additional references

Bagshaw 2007

Bagshaw SM, George C, Bellomo R, ANZICS Database Management Committee. Changes in the incidence and outcome for early acute kidney injury in a cohort of Australian intensive care units. *Critical Care (London, England)* 2007;**11**(3):R68. [MEDLINE: 17588270]

Bellomo 2013

Bellomo R, Lipcsey M, Calzavacca P, Haase M, Haase-Fielitz A, Licari E, et al. Early acid-base and blood pressure effects of continuous renal replacement therapy intensity in patients with metabolic acidosis. *Intensive Care Medicine* 2013;**39**(3):429–36. [MEDLINE: 23306586]

Bihorac 2005

Bihorac A, Ross EA. Continuous venovenous hemofiltration with citrate-based replacement fluid: efficacy, safety, and impact on nutrition. *American Journal of Kidney Diseases* 2005;**46**(5):908–18. [MEDLINE: 16253732]

Borthwick 2013

Borthwick EM, Hill CJ, Rabindranath KS, Maxwell AP, McAuley DF, Blackwood B. High-volume haemofiltration for sepsis. *Cochrane Database of Systematic Reviews* 2013, Issue 1. [DOI: 10.1002/14651858.CD008075.pub2]

Bouchard 2009

Bouchard J, Soroko SB, Chertow GM, Himmelfarb J, Ikizler TA, Paganini EP, et al. Fluid accumulation, survival and recovery of kidney function in critically ill patients with acute kidney injury. *Kidney International* 2009;**76**(4):422-7. [MEDLINE: 19436332]

Brunnet 1999

Brunnet S, Leblanc M, Geadah D, Parent D, Courteau S, Cardinal J. Diffusive and convective solute clearances during continuous renal replacement therapy at various dialysate and ultrafiltration flow rates. *American Journal of Kidney Diseases* 1999;**34**(3):486-92. [MEDLINE: 10469859]

Clark 1999

Clark WR, Ronco C. CRRT efficiency and efficacy in relation to solute size. *Kidney International - Supplement* 1999;**56**(72):S3–7. [MEDLINE: 10560796]

Clark 2014

Clark E, Molnar AO, Joannes-Boyau O, Honoré PM, Sikora L, Bagshaw SM. High-volume hemofiltration for septic acute kidney injury: a systematic review and metaanalysis. *Critical Care (London, England)* 2014;**18**(1):R7. [MEDLINE: 24398168]

Claure-Del Granado 2011

Claure-Del Granado R, Macedo E, Chertow GM, Soroko S, Himmelfarb J, Ikizler TA, et al. Effluent volume in continuous renal replacement therapy overestimates the delivered dose of dialysis.[Erratum appears in Clin J Am Soc Nephrol. 2011 Jul;6(7):1802]. *Clinical Journal of The American Society of Nephrology: CJASN* 2011;**6**(3):467-75. [MEDLINE: 21115626]

Claure-Del Granado 2012

Claure-Del Granado R, Macedo E, Chertow GM, Soroko S, Himmelfarb J, Ikizler TA, et al. Toward the optimal dose metric in continuous renal replacement therapy. *International Journal of Artificial Organs* 2012;**35**(6): 413–24. [MEDLINE: 22466995]

Davenport 2010

Davenport A, Farrington K. Dialysis dose in acute kidney injury and chronic dialysis. *Lancet* 2010;**375**(9716):705–6. [MEDLINE: 20167358]

Fayad 2013a

Fayad AI, Buamscha DG, Ciapponi A. Timing of continuous renal replacement therapy initiation for acute kidney injury. *Cochrane Database of Systematic Reviews* 2013, Issue 6. [DOI: 10.1002/14651858.CD010612]

Foland 2004

Foland JA, Fortenberry JD, Warshaw BL, Pettignano R, Merritt RK, Heard ML, et al. Fluid overload before continuous hemofiltration and survival in critically ill children: a retrospective analysis. *Critical Care Medicine* 2004;**32**(8):1771-6. [MEDLINE: 15286557]

Gibney 2008

Gibney N, Cerda J, Davenport A, Ramirez J, Singbartl K, Leblanc M, et al. Volume management by renal replacement therapy in acute kidney injury. *International Journal of Artificial Organs* 2008;**31**(2):145–55. [MEDLINE: 18311730]

Goldstein 2001

Goldstein S, Currier H, Graf JM, Cosio CC, Brewer ED, Sachdeva R. Outcome in children receiving continuous venovenous hemofiltration. *Pediatrics* 2001;**107**(6): 1309–12. [MEDLINE: 11389248]

GRADE 2008

Guyatt GH, Oxman AD, Vist GE, Kunz R, Falck-Ytter Y, Alonso-Coello P, et al. GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. *BMJ* 2008;**336**(7650):924–6. [MEDLINE: 18436948]

Grootendorst 1992

Grootendorst AF, van Bommel EF, van der Hoven B, van Leengoed LA, van Osta AL. High volume hemofiltration improves right ventricular function in endotoxin-induced shock in the pig. *Intensive Care Medicine* 1992;**18**(4): 235–40. [MEDLINE: 1430589]

Higgins 2003

Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *BMJ* 2003;**327** (7414):557–60. [MEDLINE: 12958120]

Higgins 2011

Higgins JP, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

Honore 2000

Honore PM, Jamez J, Wauthier M, Lee PA, Dugernier T, Pirenne B, et al. Prospective evaluation of short-term, highvolume isovolemic hemofiltration on the hemodynamic course and outcome in patients with intractable circulatory failure resulting from septic shock. *Critical Care Medicine* 2000;**28**(11):3581-7. [MEDLINE: 11098957]

Hoste 2006

Hoste EA, Clermont G, Kersten A, Venkataraman R, Angus DC, De Bacquer D, et al. RIFLE criteria for acute kidney injury are associated with hospital mortality in critically ill patients: a cohort analysis. *Critical Care (London, England)* 2006;**10**(3):R73. [MEDLINE: 16696865]

Joannidis 2005

Joannidis M, Metnitz PG. Epidemiology and natural history of acute renal failure in the ICU. *Critical Care Clinics* 2005; **21**(2):239-49. [MEDLINE: 15781160]

Jun 2010

Jun M, Hiddo J, Heerspink L, Ninomiya T, Gallagher M, Bellomo R, et al. Intensities of renal replacement therapy in acute kidney injury: a systematic review and meta-analysis. *Clinical Journal of The American Society of Nephrology: CJASN* 2010;**5**(6):956-63. [MEDLINE: 20395356]

KDIGO 2012

KDIGO. Dialysis interventions for treatment of AKI. *Kidney International - Supplement* 2012;**2**:89–115. [DOI: 10.1038/kisup.2011.35]

Lehner 2014

Lehner GF, Wiedermann CJ, Joannidis M. High-volume hemofiltration in critically ill patients: a systematic review and meta-analysis. *Minerva Anestesiologica* 2014;**80**(5): 595–609. [MEDLINE: 24292260]

Liao 2003

Liao Z, Zhang W, Hardy PA, Poh CK, Huang Z, Kraus MA, et al. Kinetic comparison of different acute dialysis therapies. *Artificial Organs* 2003;**2**7(9):802-7. [MEDLINE: 12940902]

Lowrie 1981

Lowrie EG, Laird NM, Parker TF, Sargent JA. Effect of the hemodialysis prescription of patient morbidity: report from

the National Cooperative Dialysis Study. *New England Journal of Medicine* 1981;**305**(20):1176-81. [MEDLINE: 7027040]

Lyndon 2012

Lyndon WD, Wille KM, Tolwani AJ. Solute clearance in CRRT: prescribed dose versus actual delivered dose. *Nephrology Dialysis Transplantation* 2012;**27**(3):952-6. [MEDLINE: 21896498]

Marshall 2006

Marshall MR. Current status of dosing and quantification of acute renal replacement therapy. Part 2: dosing paradigms and clinical implementation. *Nephrology* 2006;**11**(3): 181–91. [MEDLINE: 16756629]

Mehta 1999

Mehta RL, Letteri JM. Current status of renal replacement therapy for acute renal failure. A survey of US nephrologists. The National Kidney Foundation Council on Dialysis. *American Journal of Nephrology* 1999;**19**(3):377-82. [MEDLINE: 10393374]

Mehta 2007

Mehta RL, Kellum JA, Shah SV, Molitoris BA, Ronco C, Warnock DG, et al. Acute Kidney Injury Network: report of an initiative to improve outcomes in acute kidney injury. *Critical Care (London, England)* 2007;1(2):R31. [MEDLINE: 17331245]

Metnitz 2002

Metnitz PG, Krenn CG, Steltzer H, Lang T, Ploder J, Lenz K, et al. Effect of acute renal failure requiring renal replacement therapy on outcome in critically ill patients. *Critical Care Medicine* 2002;**30**(9):2051-8. [MEDLINE: 12352040]

Morimatsu 2003

Morimatsu H, Uchino S, Bellomo R, Ronco C. Continuous renal replacement therapy: does technique influence electrolyte and bicarbonate control?. *International Journal* of Artificial Organs 2003;**26**(4):289–96. [MEDLINE: 12757027]

Negash 2011

Negash DT, Dhingra VK, Copland M, Griesdale D, Henderson W. Intensity of continuous renal replacement therapy in acute kidney injury in the intensive care unit: a systematic review and meta-analysis. *Vascular & Endovascular Surgery* 2011;**45**(6):504–10. [MEDLINE: 21646231]

Paganini 1996

Paganini EP, Sandy D, Moreno L, Kozlowski L, Sakai K. The effect of sodium and ultrafiltration modelling on plasma volume changes and haemodynamic stability in intensive care patients receiving haemodialysis for acute renal failure: a prospective, stratified, randomized, crossover study. *Nephrology Dialysis Transplantation* 1996;**11 Suppl 8:32–7**. [MEDLINE: 9044338]

Palevsky 2002

Palevsky PM, Bunchman T, Tetta C. The Acute Dialysis Quality Initiative--part V: operational characteristics of CRRT. Advances in Renal Replacement Therapy 2002;9(4): 268-72. [MEDLINE: 12382230]

Palevsky 2005

Palevsky PM, Baldwin I, Davenport A, Goldstein S, Paganini E. Renal replacement therapy and the kidney: minimizing the impact of renal replacement therapy on recovery of acute renal failure. *Current Opinion in Critical Care* 2005;**11**(6):548–54. [MEDLINE: 16292058]

Palevsky 2009

Palevsky PM, O' Connor TZ, Chertow GM, Crowley ST, Zhang JH, Kellum JA, et al. Intensity of renal replacement therapy in acute kidney injury: perspective from within the acute Renal Failure Trial Network Study. *Critical Care (London, England)* 2009;**13**(4):310. [MEDLINE: 19678919]

Parker 1994

Parker TF 3rd, Husni L, Huang W, Lew N, Lowrie EG. Survival of hemodialysis patients in the United States is improved with a greater quantity of dialysis. *American Journal of Kidney Diseases* 1994;**23**(5):670-80. [MEDLINE: 8172209]

Phu 2002

Phu NH, Hien TT, Mai NT, Chau TT, Chuong LV, Loc PP, et al. Hemofiltration and peritoneal dialysis in infectionassociated acute renal failure in Vietnam. *New England Journal of Medicine* 2002;**347**(12):895-902. [MEDLINE: 12239258]

Ronco 2002

Ronco C, Brendolan A, Lonnemann G, Bellomo R, Piccinni P, Digito A, et al. A pilot study of coupled plasma filtration with adsorption in septic shock. *Critical Care Medicine* 2002;**30**(6):1250-5. [MEDLINE: 12072677]

Schneider 2010

Schneider J, Khemani R, Grushkin C, Bart R. Serum creatinine as stratified in the RIFLE score for acute kidney injury is associated with mortality and length of stay for children in the pediatric intensive care unit. *Critical Care Medicine* 2010;**38**(3):933-9. [MEDLINE: 20124891]

Schünemann 2011a

Schünemann HJ, Oxman AD, Higgins JP, Vist GE, Glasziou P, Guyatt GH. Chapter 11: Presenting results and 'Summary of findings' tables. In: Higgins JP, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

Schünemann 2011b

Schünemann HJ, Oxman AD, Higgins JP, Deeks JJ, Glasziou P, Guyatt GH. Chapter 12: Interpreting results and drawing conclusions. In: Higgins JP, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

Sieberth 1995

Sieberth HG, Stummvoll HK, Kierdoff H (editors). Continuous extracorporeal treatment in multiple organ dysfunction syndrome: 3rd International Conference on Continuous Hemofiltration. Vienna, July 8, 1994 (Contributions to Nephrology). Vol. **116**, Basal: Karger, 1995.

Sutherland 2010

Sutherland SM, Zappitelli M, Alexander SR, Chua AN, Brophy PD, Bunchman TE, et al. Fluid overload and mortality in children receiving continuous renal replacement therapy: the prospective pediatric continuous renal replacement therapy registry. *American Journal of Kidney Diseases* 2010;**55**(2):316-25. [MEDLINE: 20042260]

Uchino 2001

Uchino S, Bellomo R, Ronco C. Intermittent versus continuous renal replacement therapy in the ICU: impact on electrolyte and acid-base balance. *Intensive Care Medicine* 2001;**27**(6):1031–47. [MEDLINE: 11497136]

Uchino 2005

Uchino S, Kellum JA, Bellomo R, Doig GS, Morimatsu H, Morgera S, et al. Acute renal failure in critically ill patients: a multinational, multicenter study. *JAMA* 2005;**294**(7): 813–8. [MEDLINE: 16106006]

Van Wert 2010

Van Wert R, Friedrich JO, Scales DC, Wald R, Adhikari NK, University of Toronto Acute Kidney Injury Research Group. High-dose renal replacement therapy for acute kidney injury: systematic review and meta-analysis. *Critical Care Medicine* 2010;**38**(5):1360–9. [MEDLINE: 20308884]

References to other published versions of this review

Fayad 2013b

Fayad AI, Buamscha DG, Ciapponi A. Intensity of continuous renal replacement therapy for acute kidney injury. *Cochrane Database of Systematic Reviews* 2013, Issue
[DOI: 10.1002/14651858.CD010613]

* Indicates the major publication for the study

CHARACTERISTICS OF STUDIES

Characteristics of included studies [ordered by study ID]

ATN Study 2005

| Methods | Study design: prospective, parallel RCTStudy duration: November 2003 to July 2007 |
|---------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Participants | Setting: multicentre (27 ICU) Country: USA Critically ill patients aged ≥ 18 years with AKI due to ATN who require RRT defined as a) Clinical setting of Ischaemic or nephrotoxic injury, b) oliguria (average urine output ≤ 20 mL/hour) for > 24 hours or an increase in SCr of ≥ 2mg/dL (177 µmol/L) in males or ≥ 1.5 mg/dL (133 µmol/L) in females over a period of ≤ 4 days; receiving care in a critical care unit); 1 non-renal organ failure (SOFA score ≥ 2) or the presence of sepsis; patient/surrogate willing to provide informed consent Number: treatment group (563); control group (561) CRRT treatment: 783 patients (69.7%) Mean age ± SD (years): treatment group (59.6 ±15.3); control group (59.7 ± 15.2) Sex (M/F): treatment group (409/154); control group (384/176) Exclusion criteria: baseline SCr > 2 mg/dL (177 µmol/L) in males, > 1.5 mg/dL (133 µmol/L) in females; AKI clinically believed to be due to an aetiology other than ATN; more than 72 hours since meeting both of the following conditions a) fulfilment of the definition of AKI, b) BUN > 100 mg/dL (36 mmol/L); ≥ 1 HD treatment or more than 24 hours since starting CRRT; prior kidney transplant; pregnancy; prisoner; weight > 128.5 kg; non-candidacy for RRT; moribund state; patient not expected to survive 28 days because of underlying terminal chronic medical condition; comfortmeasures-only status; participation in a concurrent interventional study; patient/ surrogate refusal; physician refusal |
| Interventions | Modalities: IHD, CVVHDF, SLED Haemofilter: cellulose triacetate or synthetic membranes Replacement fluid: pre-dilution mode Anticoagulation: heparin, citrate, other Treatment group Intensive management strategy If haemodynamically stable IHD 6 times/week (target delivered Kt/V ~ 1.2 to 1.4/treatment) If haemodynamically unstable CVVHDF at 35 mL/kg/h; or SLED, 6 times/week (target delivered Kt/V ~ 1.2 to 1.4/treatment) Control group Conventional management strategy If haemodynamically stable Kt/V ~ 1.2 to 1.4/treatment) Control group Conventional management strategy If haemodynamically stable SLED, 6 times/week (target delivered Kt/V ~ 1.2 to 1.4/treatment) Control group Conventional management strategy If haemodynamically unstable SLED, 3 times/week (target delivered Kt/V ~ 1.2 to 1.4/treatment) Co-interventions Not reported |

ATN Study 2005 (Continued)

| Outcomes | Primary outcomes • Mortality from any cause at day 60 Secondary outcomes • Hospital mortality • 1 year mortality • 1 year mortality • Recovery of kidney function by day 28 Tertiary endpoints • Duration of RRT • ICU length of stay • Hospital length of stay • Discharge to "home" off of dialysis by day 60 • SOFA scores at days 1 to 14, 21 and 28 Economic analysis • RRT-specific cost of care • Global cost of care • Patient utility |
|----------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|----------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|

Notes

Risk of bias

| Bias | Authors' judgement | Support for judgement |
|------------------------------------------------------------------------------|--------------------|----------------------------------------------------------------------------------------------------------------------------|
| Random sequence generation (selection bias) | Low risk | Patients randomly assigned to one of the two treatment groups by means of a cen- tralized, computer-generated method |
| Allocation concealment (selection bias) | Low risk | Central allocation process |
| Blinding of participants and personnel (performance bias) All outcomes | Unclear risk | Insufficient information to permit judge- ment (for kidney recovery was unclear risk but for mortality was low risk) |
| Blinding of outcome assessment (detection bias) All outcomes | Low risk | The outcome measurement is not likely to be influenced by lack of blinding |
| Incomplete outcome data (attrition bias) All outcomes | Low risk | Per cent followed: 99.55% |
| Selective reporting (reporting bias) | Low risk | The study reported mortality, kidney func- tion recovery and adverse events |
| Other bias | Low risk | Funding sources were reported |

Bouman 2002

| Methods | Study design: parallel RCTStudy duration: May 1998 to March 2000 |
|---------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Participants | Setting: 2 centres Country: The Netherlands Patients with circulatory and respiratory insufficiency and early AKI who need CRRT; CrCl < 20 mL/min, and oliguria < 180 mL/6 h despite fluid resuscitation; circulatory support and furosemide; early timing: < 12 h inclusion; late timing: BUN > 40 mmol/L or severe pulmonary oedema Number: treatment group 1 (35); treatment group 2 (35); control group (36) Mean age ± SD (years): treatment group 1 (68 ± 13); treatment group 2 (70 ± 10) ; control group (67 ± 13) Sex (M/F): treatment group 1 (21/14); treatment group 2 (20/15); control group (23/13) Exclusion criteria: pre-existing kidney disease with CrCl of 30 mL/min; AKI caused by permanent occlusion or surgical lesion of the renal artery; glomerulonephritis, interstitial nephritis, or vasculitis; postrenal obstruction; CHILD class C liver cirrhosis; AIDS with a CD4 count of 0.05 109/L; non-witnessed arrest with Glasgow Coma Score 5; haematologic malignancy with neutrophil 0.05 x 10⁹/L; no haemofiltration machine free for use at the moment of inclusion |
| Interventions | Modality: CVVH Haemofilter: cellulose triacetate hollow-fibre Replacement fluid: post-dilution mode with bicarbonate solution Anticoagulation: heparin or nadroparin Treatment group 1 Early + high volume haemofiltration group Treatment started within 12 h after time of inclusion, and the ultrafiltration flow rate was high (prescribed dose > 72 L/d and delivered dose 48.2 mL/kg/h) Treatment group 2 Early + low-volume haemofiltration group Treatment started within 12 h after time of inclusion, and the ultrafiltration flow rate was low (prescribed dose 24 to 36 L/d and delivered dose 19 to 20 mL/kg/h) Control group Treatment started when the patient fulfilled the conventional criteria for RRT: urea level 40 mmol/L, potassium 6.5 mmol/L or severe pulmonary oedema, and the ultrafiltration flow rate was low (24 to 36 L/d and delivered dose 19 to 20 mL/kg/h) |
| Outcomes | Primary outcomes • Mortality at day 28 • Recovery of kidney function Secondary outcomes • ICU survival • Hospital survival • Duration of mechanical ventilation • Length of ICU stay • Length of hospitalisation |

Notes

Risk of bias

| Bias | Authors' judgement | Support for judgement |
|------------------------------------------------------------------------------|--------------------|----------------------------------------------------------------------------------------------------------------------------|
| Random sequence generation (selection bias) | Low risk | Patients randomly assigned to the treat- ment dosage using computer-generated method |
| Allocation concealment (selection bias) | Low risk | Treatment assignments were kept in num- bered, sealed opaque envelopes that were opened at the time of enrolment |
| Blinding of participants and personnel (performance bias) All outcomes | Unclear risk | Insufficient information to permit judge- ment (for kidney recovery was unclear risk but for mortality was low risk) |
| Blinding of outcome assessment (detection bias) All outcomes | Low risk | The outcome measurement is not likely to be influenced by lack of blinding |
| Incomplete outcome data (attrition bias) All outcomes | Low risk | No incomplete data were reported |
| Selective reporting (reporting bias) | Low risk | The study reported mortality, kidney func- tion recovery and adverse events |
| Other bias | Unclear risk | Insufficient information to permit judge- ment |

RENAL Study 2006

| Methods | Study design: prospective, parallel RCTStudy duration: December 2005 to November 2008 |
|--------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Participants | Setting: 35 ICU Countries: Australia and New Zealand Critically ill AKI patients who need CRRT; oliguria (urine output < 100 mL in 6 h period) unresponsive to fluid resuscitation measures; potassium > 6.5 mmol; severe acidaemia pH < 7.2/L; urea nitrogen level > 70 mg/dL or 25 mmol/L; SCr > 3.4 mg/dL or 300 µmol/L; pulmonary oedema Number: treatment group (722); control group (743) Mean age ± SD (years): treatment group (64.7 ± 14.5); control group (64.4 ± 15.3) Sex (M/F): treatment group (474/248); control group (472/271) Exclusion criteria: patients who had received any previous RRT during the same hospital admission or maintenance dialysis for study; patients with ESKD |

RENAL Study 2006 (Continued)

| Interventions | Modality: CVVHDF Qb: 150 mL/min Haemofilter: polyacrylonitrile hollow-fibre Replacement fluid: post-dilution mode with bicarbonate solution Treatment group Higher intensity CRRT Prescribed dose: 40 mL/kg/h of effluent dose (delivered dose 33.4 ± 12.8 mL/kg/h) Control group Lower intensity CRRT Prescribed dose: 25 mL/kg/h of effluent dose (delivered dose 22 ± 17.8 mL/kg/h) Co-interventions Not reported |
|---------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Outcomes | Primary outcomes Mortality at 90 days Secondary outcomes Mortality at 28 days; in the ICU; in the hospital Numbers of days of RRT; in ICU; in hospital; of mechanical ventilation Number of non-renal organ failures |

Notes

Risk of bias

| Bias | Authors' judgement | Support for judgement |
|------------------------------------------------------------------------------|--------------------|--------------------------------------------------------------------------------|
| Random sequence generation (selection bias) | Low risk | Patients were randomly assigned using computer-generated methodology |
| Allocation concealment (selection bias) | Low risk | Central allocation process |
| Blinding of participants and personnel (performance bias) All outcomes | Unclear risk | Insufficient information to permit judge- ment |
| Blinding of outcome assessment (detection bias) All outcomes | Low risk | The outcome measurement was unlikely to be influenced by lack of blinding |
| Incomplete outcome data (attrition bias) All outcomes | Low risk | Complete follow up data |
| Selective reporting (reporting bias) | Low risk | The study reported mortality, kidney func- tion recovery and adverse events |
| Other bias | Low risk | Funding sources were reported |

Ronco 2000a

| Methods | Study design: prospective, parallel RCTStudy duration: 1994 to September 1999 | |
|---------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------|
| Participants | Setting: 2 different ICU in the same hospital Country: Italy Critically ill AKI patients who need CRRT; admission to ICU Number: treatment group 1 (139); treatment group 2 (140); control group (146) Mean age ± SD (years): treatment group 1 (59 ± 9); treatment group 2 (63 ± 12); control group (61 ± 10) Sex (M/F): treatment group 1 (77/62); treatment group 2 (80/60); control group (81/65) Exclusion criteria: not reported | |
| Interventions | Modality: CVVH Haemofilters: polysulfone hollow-fibre Qb: 120 and 240 mL/min Anticoagulation: systemic heparin Replacement fluid: post-dilution mode with lactate solution Treatment group 1 Higher intensity CRRT Prescribed dose: 35 mL/kg/h (delivered dose 33.5 mL/kg/h) Treatment group 2 Higher intensity CRRT Prescribed dose: 45 mL/kg/h (delivered dose 42.5 mL/kg/h) Control group Lower intensity CRRT Prescribed dose: 20 mL/kg/h (delivered dose 19 mL/kg/h) | |
| Outcomes | Primary outcome • Survival at day 15 after discontinuation of CVVH Secondary outcomes • Recovery of kidney function 15 days after discontinuation of CVVH • Adverse events: clinical and technical complications | |
| Notes | | |
| Risk of bias | | |
| Bias | Authors' judgement | Support for judgement |

| Bias | Authors' judgement | Support for judgement |
|---------------------------------------------|--------------------|----------------------------------------------------------------------|
| Random sequence generation (selection bias) | Low risk | Patients were randomly assigned using computer-generated methodology |
| Allocation concealment (selection bias) | Low risk | Central allocation process |

Ronco 2000a (Continued)

| Blinding of participants and personnel (performance bias) All outcomes | Unclear risk | Insufficient information to permit judge- ment |
|------------------------------------------------------------------------------|--------------|------------------------------------------------------------------------|
| Blinding of outcome assessment (detection bias) All outcomes | Low risk | Outcome measurement unlikely to be in- fluenced by lack of blinding |
| Incomplete outcome data (attrition bias) All outcomes | Low risk | No losses to follow up |
| Selective reporting (reporting bias) | Low risk | Reported mortality, kidney function recov- ery and adverse events |
| Other bias | Unclear risk | Insufficient information to permit judge- ment |

Saudan 2006

| Methods | Study design: prospective, parallel RCTTime frame: October 2000 to December 2003 |
|---------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Participants | Setting: medical and surgical ICU, University Hospital Country: Switzerland Critically ill patients with AKI who need CRRT; AKI cause was mostly medical Number: treatment group (102); control group (104) Mean age ± SD (years): treatment group (65 ± 12); control group (62 ± 15) Sex (M/F): treatment group (65/37); control group (57/47) Exclusion criteria: Pre-renal failure; post-renal failure; ESKD; patients on ACEi |
| Interventions | Modalities: CVVHDF and CVVH Haemofilters: polyacrylonitrile hollow-fibre Qb: 100 to 125 mL/min Replacement fluid: pre-dilution mode with bicarbonate or lactate solution Anticoagulation: heparin Treatment group Higher intensity CRRT CVVHDF Prescribed dose: 42 mL/kg/h Control group Lower intensity CRRT CVVH Prescribed dose: 25 mL/kg/h |
| Outcomes | Primary outcome • Survival at days 28 and 90 Secondary outcomes • Recovery of kidney function |

Saudan 2006 (Continued)

| | • Length of ICU stay | |
|------------------------------------------------------------------------------|----------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Notes | | |
| Risk of bias | | |
| Bias | Authors' judgement | Support for judgement |
| Random sequence generation (selection bias) | Low risk | Patients were randomly assigned us- ing computer-generated methodology in blocks of four and six patients |
| Allocation concealment (selection bias) | Unclear risk | Allocation was appropriate (sealed opaque envelopes). However, the table 1 showed a significant imbalance in the severity of illness observed between treatment arms |
| Blinding of participants and personnel (performance bias) All outcomes | Unclear risk | Insufficient information to permit judge- ment |
| Blinding of outcome assessment (detection bias) All outcomes | Low risk | Outcome measurement unlikely to be in- fluenced by lack of blinding |
| Incomplete outcome data (attrition bias) All outcomes | Low risk | No dropouts or losses to follow up |
| Selective reporting (reporting bias) | Low risk | Reported mortality, kidney function recov- ery and adverse events |
| Other bias | Unclear risk | Insufficient information to permit judge- ment |

Tolwani 2008

| Methods | Study design: prospective, parallel RCTStudy duration: August 2003 to March 2006 |
|--------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Participants | Setting: University of Alabama (ICU) Country: USA ≥ 18 years; critically ill patients with AKI who need CRRT; AKI was mostly medical-volume overload despite diuretics; oliguria (urine output < 200 mL/12 h) despite fluid resuscitation and diuretics; anuria (urine output < 50 mL/12 h); azotaemia (BUN ≥ 80 mg/dL); hyperkalaemia (K ≥ 6.5 mmol/L); SCr increase > 2.5 mg/dL from normal values or a sustained rise in SCr ≥ 1 mg/dL over baseline Number: treatment group (100); control group (100) Mean age ± SD (years): treatment group (56 ± 16); control group (62 ± 15) Sex (M/F): treatment group (59/41); control group (57/43) |

Tolwani 2008 (Continued)

| | Relevant health status: critically ill patients with AKI who need CRRT Exclusion criteria: previous IHD treatment; 24 h CRRT at time of enrolment; weighed > 125 or < 50 kg (limitations of machine); SKD |
|---------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Interventions | Modality: CVVHDF Haemofilters: polyacrylonitrile hollow-fibre Qb: 100 to 125 mL/min Replacement fluid: pre-dilution mode Anticoagulation: heparin or no anticoagulation Treatment group Higher intensity CRRT Prescribed dose: 35 mL/kg/h Control group Lower intensity CRRT Prescribed dose: 25 mL/kg/h Co-interventions Not reported |
| Outcomes | Primary outcome • Survival in ICU discharge or 30 day Secondary outcomes • ICU survival • Hospital survival • ICU kidney recovery • Hospital kidney recovery • ICU length of stay • Hospital length of stay |

Notes

Risk of bias

| Bias | Authors' judgement | Support for judgement |
|------------------------------------------------------------------------------|--------------------|--------------------------------------------------------------------------------------------------------------------|
| Random sequence generation (selection bias) | Low risk | Patients were randomly assigned using computer-generated methodology (1:1 ra- tio between treatment dosages) |
| Allocation concealment (selection bias) | Low risk | Treatment assignments were kept in num- bered, sealed envelopes that were opened at the time of enrolment |
| Blinding of participants and personnel (performance bias) All outcomes | Unclear risk | Insufficient information to permit judge- ment |
| Blinding of outcome assessment (detection bias) All outcomes | Low risk | Outcome measurement unlikely to be in- fluenced by lack of blinding |

Tolwani 2008 (Continued)

| Incomplete outcome data (attrition bias) All outcomes | Low risk | Per cent followed: 100% |
|----------------------------------------------------------|----------|----------------------------------------------------------------------|
| Selective reporting (reporting bias) | Low risk | Reported mortality, kidney function recov- ery and adverse events |
| Other bias | Low risk | Funding sources were reported |

ACEi - angiotensin-converting enzyme inhibitors; AIDs - acquired immune deficiency syndrome; AKI - acute kidney injury; ATN - acute tubular necrosis; BUN - blood urea nitrogen; CrCl - creatinine clearance; CRRT - continuous renal replacement therapy; CVVH - continuous venovenous haemofiltration; CVVHDF - continuous venovenous haemodiafiltration; HD - haemodialysis; ICU - intensive care unit/s; IHD - intermittent haemodialysis; M/F - male/female; Qb - extracorporeal blood flow; RCT - randomised controlled trial; RRT - renal replacement therapy; SCr - serum creatinine; SD - standard deviation; SLED - sustained low-efficiency dialysis; SOFA - Sequential Organ Failure Assessment

Characteristics of excluded studies [ordered by study ID]

| Study | Reason for exclusion |
|--------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Boussekey 2008 | Different intensity-arms treatment; control arm was not within the pre-specified range according to pro- tocol: low volume HF (35 ml/kg/h) versus high volume HF (65 ml/kg/h); small sample size (< 20 partici- pants) |
| Brause 2003 | Not RCT |
| Cole 2002 | Compared continuous dialysis therapy versus no haemofiltration |
| Ghani 2006 | Outcomes not relevant for this review; intensity of CRRT was not assessed |
| HEROICS Study 2015 | 36% of patients in the control arm did not receive CRRT |
| IVOIRE Study 2013 | Different intensity arms treatment; control arm is not within the pre-specified range according to protocol: standard volume HF (35 mL/kg/h) versus high volume HF (70 mL/kg/h) |
| Jiang 2005 | Different inclusion criteria; included patients with severe pancreatitis, but AKI was no obligatory condition for enrolment; AKI was observed in only 6 (16%) patients |
| NCT01191905 | Different intensity arms treatment; less intensive arm is not within the pre-specified range according to protocol: standard volume HF (40 mL/kg/h) versus high volume HF (80 mL/kg/h) |
| NCT01251081 | Different intensity arms treatment; the control arm is not within the pre-specified range according to protocol: high volume HF (50 mL/kg/h) versus extra high volume HF (85 mL/kg/h) |
| Payen 2009 | Outcomes not relevant for this review; intensity of CRRT was not assessed |

| Sanchez 2010b | Different intensity arms treatment; control dose arms are not within the pre-specified range according to protocol: high volume HF (35 mL/kg/h) versus very high volume HF (> 55 mL/kg/h) |
|---------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Vesconi 2009 | Compared CRRT versus IHD |
| Zha 2012 | Not RCT |
| Zhang 2012 | Different intensity arms treatment; the control arm is not within the pre-specified range according to protocol: high volume HF (50 mL/kg/h) versus extra high volume HF (85 mL/Kg/h) |

CRRT - continuous renal replacement therapy; HF - haemofiltration; IHD - intermittent haemodialysis; RCT - randomised controlled trial

Characteristics of studies awaiting assessment [ordered by study ID]

NCT01560650

| Methods | Study design: RCT Study duration: March 2011 to August 2015 (final data collection date for primary outcome measure) |
|---------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Participants | Setting: Guangdong General Hospital Country: China CRRT indications for AKI (RIFLE criteria) patients with cardiac surgery Number: 211 Age: ≥ 18 years Exclusion criteria: < 18 years; CKD; dialysis history, to leave the ICU patients with AKI, CKD; all causes kidney damage (pathology, haematuria, and radiographic abnormalities) ≥ 3 months or GFR < 60 mL/min for 3 months or more |
| Interventions | Treatment group • High dose (35 mL/kg/h) CVVH Control group • Low dose (25 mL/kg/h) CVVH |
| Outcomes | Primary outcomes Death from any cause within 14, 28 and 90 days after randomisation Secondary outcomes Kidney outcome of survivors 14, 28 and 90 days after randomisation |
| Notes | • This study is now completed; no study results have been posted (September 2016) |

AKI - acute kidney injury; CKD - chronic kidney disease; CRRT - continuous renal replacement therapy; CVVH - continuous venovenous haemofiltration; GFR - glomerular filtration rate; ICU - intensive care unit/s; RCT - randomised controlled trial

DATA AND ANALYSES

Comparison 1. Intensive versus less intensive CRRT

| Outcome or subgroup title | No. of studies | No. of participants | Statistical method | Effect size |
|-------------------------------------------------------|-------------------|------------------------|---------------------------------------|---------------------|
| 1 Mortality | 6 | | Risk Ratio (M-H, Random, 95% CI) | Subtotals only |
| 1.1 Mortality at day 30 | 5 | 2402 | Risk Ratio (M-H, Random, 95% CI) | 0.88 [0.71, 1.08] |
| 1.2 Mortality after 30 days | 5 | 2759 | Risk Ratio (M-H, Random, 95% CI) | 0.92 [0.80, 1.06] |
| post-randomisation | | | | |
| 2 Mortality in prespecified groups | 5 | | Risk Ratio (M-H, Random, 95% CI) | Subtotals only |
| 2.1 Patients with sepsis | 5 | 966 | Risk Ratio (M-H, Random, 95% CI) | 0.94 [0.69, 1.27] |
| 2.2 Patients without sepsis | 4 | 1216 | Risk Ratio (M-H, Random, 95% CI) | 0.89 [0.69, 1.15] |
| 2.3 Patients with SOFA | 1 | 404 | Risk Ratio (M-H, Random, 95% CI) | 0.91 [0.71, 1.18] |
| cardiovascular score < 3 | | | | |
| 2.4 Patients with SOFA | 1 | 1056 | Risk Ratio (M-H, Random, 95% CI) | 1.04 [0.92, 1.18] |
| cardiovascular ≥ 3 | | | | |
| 2.5 Patients with AKI related | 2 | 531 | Risk Ratio (M-H, Random, 95% CI) | 0.73 [0.61, 0.88] |
| 2.6 Patients with AKI | 3 | 1871 | Risk Ratio (M-H, Random, 95% CI) | 0.94 [0.73, 1.20] |
| 3 Recovery of kidney function | 5 | | Risk Ratio (M-H. Random, 95% CI) | Subtotals only |
| 3.1 Free of RRT after | 5 | 2/02 | Risk Ratio (M H Random, 95% CI) | |
| discontinuing CRRT |) | 2402 | Risk Ratio (IVI-11, Randoni, 9970 CI) | 1.12 [0.91, 1.97] |
| 3.2 Free of RRT after discontinuing CRRT at day 30 | 5 | 1416 | Risk Ratio (M-H, Random, 95% CI) | 1.03 [0.96, 1.11] |
| 3.3 Free of RRT after | 3 | 988 | Risk Ratio (M-H, Random, 95% CI) | 0.98 [0.94, 1.01] |
| discontinuing CRR1 at day 90 | ~ | | | 0 1 1 1 |
| 4 Kidney function recovery in prespecified subgroup |) | | Risk Ratio (M-H, Random, 95% CI) | Subtotals only |
| 4.1 Patients with AKI related to surgical causes | 2 | 531 | Risk Ratio (M-H, Random, 95% CI) | 1.27 [1.05, 1.53] |
| 4.2 Patients with AKI related to non-surgical causes | 3 | 1870 | Risk Ratio (M-H, Random, 95% CI) | 1.12 [0.73, 1.71] |
| 5 Length of stay | 2 | | Mean Difference (IV, Random, 95% CI) | Subtotals only |
| 5.1 Days in hospital | 2 | 1665 | Mean Difference (IV, Random, 95% CI) | -0.23 [-3.35, 2.89] |
| 5.2 Days in ICU | 2 | 1665 | Mean Difference (IV, Random, 95% CI) | -0.58 [-3.73, 2.56] |
| 6 Metabolic control | 1 | | Risk Ratio (M-H, Random, 95% CI) | Totals not selected |
| 6.1 Normalised metabolic acidosis | 1 | | Risk Ratio (M-H, Random, 95% CI) | 0.0 [0.0, 0.0] |
| 7 Adverse events | 3 | | Risk Ratio (M-H, Random, 95% CI) | Subtotals only |
| 7.1 Patients experiencing | 3 | 1753 | Risk Ratio (M-H. Random, 95% CI) | 1.08 [0.73, 1.61] |
| adverse events | 5 | -1.20 | | 1.00 [0., 0, 1.01] |
| 7.2 Hypophosphataemia | 1 | 1441 | Risk Ratio (M-H, Random, 95% CI) | 1.21 [1.11, 1.31] |
| 7.3 Hypokalaemia | 1 | 1455 | Risk Ratio (M-H, Random, 95% CI) | 0.96 [0.80, 1.15] |
| 7.4 Arrhythmia | 1 | 1463 | Risk Ratio (M-H, Random, 95% CI) | 0.92 [0.80, 1.06] |
| 7.5 Bleeding | 3 | 1775 | Risk Ratio (M-H, Random, 95% CI) | 0.78 [0.27, 2.24] |

Analysis I.I. Comparison I Intensive versus less intensive CRRT, Outcome I Mortality.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT

Outcome: I Mortality

| Study or subgroup | Intensive | Less intensive | Risk Ratio | Weight | Risk Ratio |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|--------------------|--------------------------|---------------------------------------------------|
| | n/N | n/N | H,Random,95% Cl | | H,Random,95% |
| Mortality at day 30 | | | | | |
| Saudan 2006 | 43/104 | 62/102 | | 19.7 % | 0.68 [0.52, 0.90] |
| Ronco 2000a | 119/279 | 86/146 | | 23.8 % | 0.72 [0.60, 0.88] |
| Bouman 2002 | 10/35 | 23/71 | | 8.4 % | 0.88 [0.47, 1.64] |
| RENAL Study 2006 | 278/722 | 274/743 | + | 26.6 % | 1.04 [0.92, 1.19] |
| Tolwani 2008 | 60/100 | 55/100 | - | 21.5 % | 1.09 [0.86, 1.39] |
| Subtotal (95% CI) | 1240 | 1162 | • | 100.0 % | 0.88 [0.71, 1.08] |
| Total events: 510 (Intensive), | 500 (Less intensive) | | | | |
| Heterogeneity: $Tau^2 = 0.04$; | Chi ² = 16.14, df = 4 | $(P = 0.003); I^2 = 75\%$ | | | |
| Test for overall effect: $Z = 1.2$ | 23 (P = 0.22) | | | | |
| 2 Mortality after 30 days pos | t-randomisation | | | | |
| Saudan 2006 | 43/104 | 67/102 | | 15.9 % | 0.63 [0.48, 0.82] |
| Bouman 2002 | I 3/35 | 32/71 | | 6.7 % | 0.82 [0.50, 1.36] |
| ATN Study 2005 | 229/397 | 229/386 | + | 28.7 % | 0.97 [0.86, 1.09] |
| RENIAL Study 2006 | 322/721 | 332/743 | + | 29.1 % | 1.00 [0.89, 1.12] |
| THE WILL STUDY 2000 | | | | | |
| Tolwani 2008 | 64/100 | 60/100 | - | 19.6 % | 1.07 [0.86, 1.33] |
| Tolwani 2008 Subtotal (95% CI) | 64/100 1357 | 60/100 1402 | • | 19.6 % 100.0 % | 1.07 [0.86, 1.33] 0.92 [0.80, 1.06] |
| Tolwani 2008 Subtotal (95% CI) Total events: 671 (Intensive), | 64/100 1357 720 (Less intensive) | 60/100 1402 | • | 19.6 % 100.0 % | 1.07 [0.86, 1.33] 0.92 [0.80, 1.06] |
| Tolwani 2008 Subtotal (95% CI) Total events: 671 (Intensive), Heterogeneity: Tau ² = 0.02; 4 | 64/100 1357 720 (Less intensive) Chi ² = 11.29, df = 4 | 60/100 1402 + (P = 0.02); I ² =65% | • | 19.6 % 100.0 % | 1.07 [0.86, 1.33] 0.92 [0.80, 1.06] |
| Tolwani 2008 Subtotal (95% CI) Total events: 671 (Intensive), Heterogeneity: Tau ² = 0.02; G Test for overall effect: Z = 1. | 64/100 1357 720 (Less intensive) Chi ² = 11.29, df = 4 12 (P = 0.26) | 60/100 1402 + (P = 0.02); I ² =65% | • | 19.6 % 100.0 % | 1.07 [0.86, 1.33] 0.92 [0.80, 1.06] |
| Tolwani 2008 Subtotal (95% CI) Total events: 671 (Intensive), Heterogeneity: Tau ² = 0.02; (Test for overall effect: $Z = I$. Test for subgroup differences | 64/100 1357 720 (Less intensive) Chi ² = 11.29, df = 4 12 (P = 0.26) : Chi ² = 0.15, df = 1 | 60/100 1402 F (P = 0.02); I ² =65% (P = 0.70), I ² =0.0% | • | 19.6 % 100.0 % | 1.07 [0.86, 1.33] 0.92 [0.80, 1.06] |

Favours intensive Favours less intensive

Analysis 1.2. Comparison I Intensive versus less intensive CRRT, Outcome 2 Mortality in prespecified groups.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT

Outcome: 2 Mortality in prespecified groups

| Study or subgroup | Intensive | Less intensive | Risk Ratio | Weight | Risk Ratio |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|---------------------------------|-------------------------------------------------------------|---------|---------------------|
| | n/N | n/N | H,Random,95% Cl | | H,Random,95% Cl |
| I Patients with sepsis | | | | | |
| Bouman 2002 | 8/14 | 4/14 | | 7.8 % | 2.00 [0.78, 5.14] |
| RENAL Study 2006 | 168/359 | 186/363 | - | 30.8 % | 0.91 [0.79, 1.06] |
| Ronco 2000a | 4/ 7 | 15/20 | + | 23.6 % | 1.10 [0.79, 1.54] |
| Saudan 2006 | 7/37 | 21/34 | _ | 11.6 % | 0.31 [0.15, 0.63] |
| Tolwani 2008 | 37/54 | 34/54 | + | 26.2 % | 1.09 [0.83, 1.43] |
| Subtotal (95% CI) | 481 | 485 | • | 100.0 % | 0.94 [0.69, 1.27] |
| Total events: 234 (Intensive), $\frac{1}{2}$ Heterogeneity: Tau ² = 0.07; $\frac{1}{2}$ Test for overall effect: Z = 0.4 | 260 (Less intensive) Chi ² = 14.33, df = 4 H3 (P = 0.67) | (P = 0.01); I ² =72% | | | |
| 2 Patients without sepsis RENAL Study 2006 | 154/362 | 145/379 | - | 29.9 % | . [0.93, .32] |
| Ronco 2000a | 46/122 | 71/126 | - | 24.7 % | 0.67 [0.51, 0.88] |
| Saudan 2006 | 36/67 | 46/68 | - | 24.6 % | 0.79 [0.60, 1.05] |
| Tolwani 2008 | 27/46 | 26/46 | - | 20.8 % | 1.04 [0.73, 1.47] |
| Subtotal (95% CI) | 59 7 | 619 | • | 100.0 % | 0.89 [0.69, 1.15] |
| Total events: 263 (Intensive), 2 Heterogeneity: Tau ² = 0.05; C Test for overall effect: $Z = 0.9$ 3 Patients with SOFA cardiov | 288 (Less intensive) Chi ² = 11.16, df = 3 90 (P = 0.37) ascular score < 3 | (P = 0.01); l ² =73% | | | |
| RENAL Study 2006 | 74/210 | 75/194 | | 100.0 % | 0.91 [0.71, 1.18] |
| Subtotal (95% CI) Total events: 74 (Intensive), 72 Heterogeneity: not applicable Test for overall effect: Z = 0.7 4 Patients with SOFA cardiov | 210 5 (Less intensive) 71 (P = 0.48) ascular > 3 | 194 | • | 100.0 % | 0.91 [0.71, 1.18] |
| RENAL Study 2006 | 247/510 | 254/546 | - | 100.0 % | 1.04 [0.92, 1.18] |
| Subtotal (95% CI) Total events: 247 (Intensive), Heterogeneity: not applicable | 510 254 (Less intensive) | 546 | · · · · · · · · · · | 100.0 % | 1.04 [0.92, 1.18] |
| | | F | 0.1 0.2 0.5 1 2 5 10 avours intensive Favours less inten | sive | |

(Continued \dots)

| Study or subgroup | Intensive | Less intensive | Risk Ratio M- H,Random,95% | Weight | (Continued) Risk Ratio M- H,Random,959 |
|------------------------------------|------------------------------------|-------------------------------|----------------------------------|---------|--------------------------------------------------|
| | n/N | n/N | ČI | | Ċl |
| Test for overall effect: $Z = 0.6$ | 52 (P = 0.53) | | | | |
| 5 Patients with AKI related to | surgical causes | | | | |
| Bouman 2002 | 10/36 | 23/70 | | 8.6 % | 0.85 [0.45, 1.58] |
| Ronco 2000a | 119/279 | 86/146 | | 91.4 % | 0.72 [0.60, 0.88] |
| Subtotal (95% CI) | 315 | 216 | • | 100.0 % | 0.73 [0.61, 0.88] |
| Total events: 129 (Intensive), | 109 (Less intensive) | | | | |
| Heterogeneity: $Tau^2 = 0.0$; Cl | $hi^2 = 0.22, df = 1$ (P | = 0.64); I ² =0.0% | | | |
| Test for overall effect: $Z = 3.3$ | 30 (P = 0.00095) | | | | |
| 6 Patients with AKI unrelated | to surgical causes | | | | |
| RENAL Study 2006 | 278/722 | 274/743 | – | 39.6 % | 1.04 [0.92, 1.19] |
| Saudan 2006 | 43/104 | 62/102 | | 28.8 % | 0.68 [0.52, 0.90] |
| Tolwani 2008 | 60/100 | 55/100 | + | 31.6 % | 1.09 [0.86, 1.39] |
| Subtotal (95% CI) | 926 | 945 | + | 100.0 % | 0.94 [0.73, 1.20] |
| Total events: 381 (Intensive), | 391 (Less intensive) | | | | |
| Heterogeneity: $Tau^2 = 0.04$; (| Chi ² = 8.32, df = 2 (I | $P = 0.02$; $I^2 = 76\%$ | | | |
| Test for overall effect: $Z = 0.5$ | 52 (P = 0.60) | | | | |
| Test for subgroup differences: | $Chi^2 = 9.56, df = 5$ | $(P = 0.09), I^2 = 48\%$ | | | |
| | | | | | |

0.1 0.2 0.5 1 2 5 10

Favours intensive Favours less intensive

Analysis 1.3. Comparison I Intensive versus less intensive CRRT, Outcome 3 Recovery of kidney function.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT

Outcome: 3 Recovery of kidney function

| Study or subgroup | Intensive | Less intensive | Risk Ratio | Weight | Risk Ratio | |
|-------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|-----------------------------------------------------------|------------------------------------|---------|---------------------|--|
| | n/N | n/N | H,Random,95% Cl | | H,Random,95% Cl | |
| I Free of RRT after discontinu | uing CRRT | | | | | |
| Bouman 2002 | 26/35 | 51/71 | | 21.5 % | 1.03 [0.81, 1.32] | |
| RENAL Study 2006 | 372/722 | 393/743 | + | 29.0 % | 0.97 [0.88, 1.07] | |
| Ronco 2000a | 145/279 | 57/146 | | 22.1 % | 1.33 [1.06, 1.68] | |
| Saudan 2006 | 48/104 | 28/102 | | 14.9 % | 1.68 [1.15, 2.45] | |
| Tolwani 2008 | 25/100 | 32/100 | | 12.5 % | 0.78 [0.50, 1.22] | |
| Subtotal (95% CI) | 1240 | 1162 | + | 100.0 % | 1.12 [0.91, 1.37] | |
| Total events: 616 (Intensive), 5 Heterogeneity: $Tau^2 = 0.03$; C Test for overall effect: $Z = 1.0$ 2 Free of BBT after discontinu | 561 (Less intensive) Chi ² = 13.97, df = 4 6 (P = 0.29) uing CBBT at day 30 | $(P = 0.01); I^2 = 71\%$ | | | | |
| Bouman 2002 | 26/26 | 51/51 | – | 28.2 % | 1.00 [0.94, 1.06] | |
| RENAL Study 2006 | 379/443 | 412/469 | - | 29.4 % | 0.97 [0.93, 1.02] | |
| Ronco 2000a | 145/160 | 57/81 | - | 14.1 % | 1.29 [1.11, 1.50] | |
| Saudan 2006 | 59/61 | 38/40 | + | 23.5 % | 1.02 [0.94, 1.11] | |
| Tolwani 2008 | 25/40 | 31/45 | | 4.8 % | 0.91 [0.67, 1.24] | |
| Subtotal (95% CI) | 730 | 686 | • | 100.0 % | 1.03 [0.96, 1.11] | |
| Total events: 634 (Intensive), 5 Heterogeneity: Tau ² = 0.00; C Test for overall effect: $Z = 0.7$ 3 Free of RRT after discontinu | 589 (Less intensive) Chi ² = 13.08, df = 4 3 (P = 0.46) uing CRRT at day 90 | (P = 0.01); I ² =69% | | | | |
| RENAL Study 2006 | 372/399 | 393/411 | - | 96.6 % | 0.98 [0.94, 1.01] | |
| Saudan 2006 | 48/62 | 28/40 | + | 1.8 % | . [0.87, .4] | |
| Tolwani 2008 | 25/36 | 32/40 | | 1.5 % | 0.87 [0.67, 1.13] | |
| Subtotal (95% CI) | 497 | 491 | • | 100.0 % | 0.98 [0.94, 1.01] | |
| Total events: 445 (Intensive), 4 Heterogeneity: Tau ² = 0.0; Ch Test for overall effect: Z = 1.4 Test for subgroup differences: | 453 (Less intensive) $h^2 = 1.78$, df = 2 (F 7 (P = 0.14) Chi ² = 3.06, df = 2 | $P = 0.41$); $I^2 = 0.0\%$: (P = 0.22), $I^2 = 35\%$ | | | | |
| | | | 0.2 0.5 2 5 | | | |
| | | Favour | s less intensive Eavours intensive | 2 | | |

Analysis I.4. Comparison I Intensive versus less intensive CRRT, Outcome 4 Kidney function recovery in prespecified subgroup.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT

Outcome: 4 Kidney function recovery in prespecified subgroup

| Study or subgroup | Intensive | Less intensive | Risk Ratio | Weight | Risk Ratio |
|------------------------------------|----------------------------------|----------------------------------|--------------------|---------|---------------------|
| | n/N | n/N | H,Random,95% Cl | | H,Random,95% Cl |
| I Patients with AKI related to | surgical causes | | | | |
| Bouman 2002 | 22/35 | 39/71 | | 33.0 % | 1.14 [0.82, 1.59] |
| Ronco 2000a | 145/279 | 57/146 | - | 67.0 % | 1.33 [1.06, 1.68] |
| Subtotal (95% CI) | 314 | 217 | • | 100.0 % | 1.27 [1.05, 1.53] |
| Total events: 167 (Intensive), | 96 (Less intensive) | | | | |
| Heterogeneity: $Tau^2 = 0.0$; Ch | $hi^2 = 0.56, df = 1$ (F | P = 0.46); I ² =0.0% | | | |
| Test for overall effect: $Z = 2.4$ | 14 (P = 0.015) | | | | |
| 2 Patients with AKI related to | non-surgical causes | | | | |
| RENAL Study 2006 | 372/721 | 393/743 | - | 40.8 % | 0.98 [0.88, 1.08] |
| Saudan 2006 | 48/104 | 25/102 | | 30.5 % | 1.88 [1.26, 2.81] |
| Tolwani 2008 | 25/100 | 32/100 | | 28.7 % | 0.78 [0.50, 1.22] |
| Subtotal (95% CI) | 925 | 945 | - | 100.0 % | 1.12 [0.73, 1.71] |
| Total events: 445 (Intensive), • | 450 (Less intensive) | | | | |
| Heterogeneity: $Tau^2 = 0.11$; (| Chi ² = 11.17, df = 2 | (P = 0.004); I ² =82% | | | |
| Test for overall effect: $Z = 0.5$ | 52 (P = 0.61) | | | | |
| Test for subgroup differences: | $Chi^2 = 0.27, df = 1$ | $(P = 0.60), ^2 = 0.0\%$ | | | |
| | | | | | |
| | | | 0.2 0.5 I 2 5 | | |

Favours less intensive Favours intensive

Analysis I.5. Comparison I Intensive versus less intensive CRRT, Outcome 5 Length of stay.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT

Outcome: 5 Length of stay

| Study or subgroup | Intensive | | Less intensive | | Mean Difference | Weight | Mean Difference |
|----------------------------|--------------------------|-------------------------|---------------------------|----------------|----------------------------|-------------|------------------------|
| | Ν | Mean(SD)[days] | Ν | Mean(SD)[days] | IV,Random,95% CI | | IV,Random,95% CI |
| I Days in hospital | | | | | | | |
| RENAL Study 2006 | 722 | 26 (25.8) | 743 | 25.7 (24.7) | + | 90.0 % | 0.30 [-2.29, 2.89] |
| Tolwani 2008 | 100 | 35 (30) | 100 | 40 (39) | | 10.0 % | -5.00 [-14.64, 4.64] |
| Subtotal (95% CI) | 822 | | 843 | | • | 100.0 % | -0.23 [-3.35, 2.89] |
| Heterogeneity: $Tau^2 = I$ | .07; $Chi^2 = 1$ | .08, df = 1 (P = 0.30 |)); l ² =8% | | | | |
| Test for overall effect: Z | = 0.14 (P = | 0.88) | | | | | |
| 2 Days in ICU | | | | | | | |
| RENAL Study 2006 | 722 | .8 (4.) | 743 | .8 (4.2) | • | 88.3 % | 0.0 [-1.45, 1.45] |
| Tolwani 2008 | 100 | 26 (26) | 100 | 31 (36) | | 11.7 % | -5.00 [-13.70, 3.70] |
| Subtotal (95% CI) | 822 | | 843 | | • | 100.0 % | -0.58 [-3.73, 2.56] |
| Heterogeneity: $Tau^2 = 2$ | .37; Chi ² = | .23, df = 1 (P = 0.27 | 7); ² = 9% | | | | |
| Test for overall effect: Z | = 0.36 (P = | 0.72) | | | | | |
| Test for subgroup differe | nces: Chi ² = | 0.02, $df = 1$ (P = 0.8 | 38), l ² =0.0% | | | | |
| | | | | | | | |
| | | | | -2 | 20 -10 0 10 | 20 | |
| | | | | Favo | ours intensive Favours les | s intensive | |

Analysis 1.6. Comparison I Intensive versus less intensive CRRT, Outcome 6 Metabolic control.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT

Outcome: 6 Metabolic control

| Study or subgroup | Intensive | Less intensive | Risk Ratio M- H,Random,95% | Risk Ratio M- H,Random,95% |
|-----------------------------------------------------|-----------|----------------|------------------------------------------|----------------------------------|
| | n/N | n/N | Cl | CI |
| I Normalised metabolic acidosis RENAL Study 2006 | 29/56 | 29/59 | | 1.05 [0.73, 1.51] |
| | | | | |
| | | | 0.5 0.7 .5 2 | |
| | | | Favours less intensive Favours intensive | |

Analysis 1.7. Comparison | Intensive versus less intensive CRRT, Outcome 7 Adverse events.

Review: Intensity of continuous renal replacement therapy for acute kidney injury

Comparison: I Intensive versus less intensive CRRT Outcome: 7 Adverse events

| Study or subgroup | Intensive | Less intensive | Risk Ratio | Weight | Risk Ratio |
|-------------------------------------------|--------------------------|----------------------------------|---------------------------------------|---------|-----------------------|
| _ | n/N | n/N | H,Random,95% Cl | | H,Random,95% Cl |
| I Patients experiencing adver | se events | | | | |
| Bouman 2002 | 4/35 | 15/71 | | 12.9 % | 0.54 [0.19, 1.51] |
| RENAL Study 2006 | 461/708 | 396/733 | - | 85.1 % | .2 [. , .3] |
| Saudan 2006 | 1/104 | 1/102 | | 2.0 % | 0.98 [0.06, 5.47] |
| Subtotal (95% CI) | 847 | 906 | • | 100.0 % | 1.08 [0.73, 1.61] |
| Total events: 466 (Intensive), | 412 (Less intensive) | | | | |
| Heterogeneity: Tau ² = 0.05; (| $Chi^2 = 2.38, df = 2$ (| P = 0.30); I ² = I 6% | | | |
| Test for overall effect: $Z = 0.3$ | 89 (P = 0.70) | | | | |
| 2 Hypophosphataemia | | | | | |
| RENAL Study 2006 | 461/708 | 396/733 | • • • • • • • • • • • • • • • • • • • | 100.0 % | .2 [. , .3] |
| Subtotal (95% CI) | 708 | 733 | ٠ | 100.0 % | 1.21 [1.11, 1.31] |
| | | | 0.005 0.1 1 10 200 | | |
| | | | Favours intensive Favours less inten | sive | |

(Continued . . .)

| Study or subgroup | Intensive | Less intensive | Risk Ratio M- | Weight | (Continued) Risk Ratio M- |
|-------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------|-------------------------------|-------------------------------------|---------|----------------------------------|
| | n/N | n/N | H,Random,95% Cl | | H,Random,95% Cl |
| Total events: 461 (Intensive), 3 | 396 (Less intensive) | | | | |
| Heterogeneity: not applicable | | | | | |
| Test for overall effect: $Z = 4.2$ | 6 (P = 0.000020) | | | | |
| 3 Hypokalaemia | | | | | |
| RENAL Study 2006 | 168/718 | 180/737 | + | 100.0 % | 0.96 [0.80, 1.15] |
| Subtotal (95% CI) | 718 | 737 | + | 100.0 % | 0.96 [0.80, 1.15] |
| Total events: 168 (Intensive), 1 Heterogeneity: not applicable Test for overall effect: $Z = 0.4$ 4 Arrhythmia | 80 (Less intensive) 6 (P = 0.65) | | | | |
| RENAL Study 2006 | 240/722 | 267/741 | + | 100.0 % | 0.92 [0.80, 1.06] |
| Subtotal (95% CI) | 722 | 741 | | 100.0 % | 0.92 [0.80, 1.06] |
| Total events: 240 (Intensive), 2 Heterogeneity: not applicable Test for overall effect: $Z = 1.1$ 5 Bleeding | 267 (Less intensive) 2 (P = 0.26) | | | | |
| Bouman 2002 | 3/35 | 10/71 | | 74.4 % | 0.61 [0.18, 2.07] |
| RENAL Study 2006 | 1/722 | 0/741 | | 10.9 % | 3.08 [0.13, 75.45] |
| Saudan 2006 | 1/104 | 1/102 | _ | 14.7 % | 0.98 [0.06, 15.47] |
| Subtotal (95% CI) | 861 | 914 | - | 100.0 % | 0.78 [0.27, 2.24] |
| Total events: 5 (Intensive), 11 Heterogeneity: $Tau^2 = 0.0$; Ch | (Less intensive) $hi^2 = 0.90$, df = 2 (P | = 0.64); l ² =0.0% | | | |
| Test for overall effect: $Z = 0.4$ | 6 (P = 0.64) | | | | |
| Test for subgroup differences: | Chi ² = 12.91, df = | 4 (P = 0.01), $I^2 = 69\%$ | | | |
| | | | 0.005 0.1 1.0 200 | | |
| | | | Favours intensive Favours less inte | nsive | |

APPENDICES

Appendix I. Electronic search strategies

| Database | Search terms |
|----------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| CENTRAL | MeSH descriptor Acute Kidney Injury explode all trees "acute kidney failure" :ti,ab,kw or "acute renal failure":ti,ab,kw in Clinical Trials "acute kidney injury" :ti,ab,kw or "acute renal injury":ti,ab,kw in Clinical Trials "acute kidney insufficiency":ti,ab,kw or "acute renal insufficiency":ti,ab,kw in Clinical Trials "acute tubular necrosis":ti in Clinical Trials (ARI or ARF or ARF or ATN):ti,ab,kw in Clinical Trials (41 OR #2 OR #3 OR #4 OR #5 OR #6) MeSH descriptor Renal Replacement Therapy, this term only MeSH descriptor Renal Dialysis explode all trees continuous NEAR/2 haemofiltration:ti,ab,kw in Clinical Trials continuous NEAR/2 haemofiltration):ti,ab,kw in Clinical Trials (continuous NEAR/2 haemofiltration):ti,ab,kw in Clinical Trials (CVVH or CVVHDF or CVVHD or SCUF or CRRT):ti,ab,kw in Clinical Trials (#8 OR #9 OR #10 OR #11 OR #12 OR #13 OR #14 OR #15 OR #16 OR #17) (#7 AND #18) |
| MEDLINE | exp Acute Kidney Injury/ (acute kidney failure or acute renal failure).tw. (acute kidney injur\$ or acute renal injur\$).tw. (acute kidney insufficie\$ or acute renal insufficie\$).tw. acute tubular necrosis.tw. (ARI or AKI or ARF or AKF or ATN).tw. or/1-6 Renal Replacement Therapy/ exp Renal Dialysis/ (continuous adj3 haemofiltration).tw. (continuous adj3 haemodialysis).tw. continuous ultrafiltration).tw. (CVVH or CVVHDF or CVVHD or SCUF or CRRT).tw. renal replacement therap\$.tw. or/8-15 and/7,16 |
| EMBASE | acute kidney failure/ (acute kidney failure or acute renal failure).tw. (acute kidney injur\$ or acute renal injur\$).tw. (acute kidney insufficie\$ or acute renal insufficie\$).tw. acute tubular necrosis.tw. (ARI or AKI or ARF or AKF or ATN).tw. or/1-6 |

| | continuous renal replacement therapy/ or exp renal replacement therapy/ (continuous adj3 hemofiltration).tw. (continuous adj3 hemodiafiltration).tw. (continuous adj3 h?emodialysis).tw. continuous ultrafiltration.tw. (CVVH or CVVHDF or CVVHD or SCUF or CRRT).tw. renal replacement therap\$.tw. or/8-14 |
|--------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| LILACS | acute kidney failure/ (acute kidney failure or acute renal failure) tw. acute tubular necrosis.tw. or/1-3 continuous renal replacement therapy/ (continuous venovenous haemofiltration or continuous venovenous hemofiltration). tw (continuous venovenous haemodiafiltration or continuous venovenous hemodiafiltration).tw. (continuous venovenous haemodiafiltration or continuous venovenous hemodiafiltration).tw. (continuous venovenous haemodialysis or continuous venovenous hemodialysis).tw. or/5-8 |

Appendix 2. Risk of bias assessment tool

| Potential source of bias | Assessment criteria | |
|-----------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| Random sequence generation Selection bias (biased allocation to interventions) due to inade- quate generation of a randomised sequence | <i>Low risk of bias:</i> Random number table; computer random number generator; coin tossing; shuffling cards or envelopes; throwing dice; drawing of lots; minimisation (minimisation may be implemented without a random element, and this is considered to be equivalent to being random) | |
| | <i>High risk of bias:</i> Sequence generated by odd or even date of birth; date (or day) of admission; sequence generated by hospital or clinic record number; allocation by judgement of the clinician; by preference of the participant; based on the results of a laboratory test or a series of tests; by availability of the intervention | |
| | <i>Unclear:</i> Insufficient information about the sequence generation process to permit judgement | |
| Allocation concealment Selection bias (biased allocation to interventions) due to inade- quate concealment of allocations prior to assignment | <i>Low risk of bias:</i> Randomisation method described that would not allow investigator/participant to know or influence intervention group before eligible participant entered in the study (e.g. central allocation, including telephone, web-based, and pharmacy-con- trolled, randomisation; sequentially numbered drug containers of identical appearance; sequentially numbered, opaque, sealed en- velopes) | |

| <i>High risk of bias:</i> Using an open random allocation schedule (e.g. a list of random numbers); assignment envelopes were used without appropriate safeguards (e.g. if envelopes were unsealed or non-opaque or not sequentially numbered); alternation or rotation; date of birth; case record number; any other explicitly unconcealed procedure |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| <i>Unclear</i> : Randomisation stated but no information on method used is available |
| <i>Low risk of bias</i> : No blinding or incomplete blinding, but the review authors judge that the outcome is not likely to be influenced by lack of blinding; blinding of participants and key study personnel ensured, and unlikely that the blinding could have been broken |
| <i>High risk of bias</i> : No blinding or incomplete blinding, and the outcome is likely to be influenced by lack of blinding; blinding of key study participants and personnel attempted, but likely that the blinding could have been broken, and the outcome is likely to be influenced by lack of blinding |
| Unclear: Insufficient information to permit judgement |
| <i>Low risk of bias:</i> No blinding of outcome assessment, but the review authors judge that the outcome measurement is not likely to be influenced by lack of blinding; blinding of outcome assessment ensured, and unlikely that the blinding could have been broken |
| <i>High risk of bias:</i> No blinding of outcome assessment, and the outcome measurement is likely to be influenced by lack of blinding; blinding of outcome assessment, but likely that the blinding could have been broken, and the outcome measurement is likely to be influenced by lack of blinding |
| Unclear: Insufficient information to permit judgement |
| Low risk of bias: No missing outcome data; reasons for missing outcome data unlikely to be related to true outcome (for survival data, censoring unlikely to be introducing bias); missing outcome data balanced in numbers across intervention groups, with similar reasons for missing data across groups; for dichotomous outcome data, the proportion of missing outcomes compared with observed event risk not enough to have a clinically relevant impact on the intervention effect estimate; for continuous outcome data, plau- sible effect size (difference in means or standardised difference in means) among missing outcomes not enough to have a clinically relevant impact on observed effect size; missing data have been |
| |

| | imputed using appropriate methods | |
|---------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| | <i>High risk of bias:</i> Reason for missing outcome data likely to be related to true outcome, with either imbalance in numbers or reasons for missing data across intervention groups; for dichotomous outcome data, the proportion of missing outcomes compared with observed event risk enough to induce clinically relevant bias in intervention effect estimate; for continuous outcome data, plausible effect size (difference in means or standardised difference in means) among missing outcomes enough to induce clinically relevant bias in observed effect size; 'as-treated' analysis done with substantial departure of the intervention received from that assigned at randomisation; potentially inappropriate application of simple imputation | |
| | Unclear: Insufficient information to permit judgement | |
| Selective reporting Reporting bias due to selective outcome reporting | <i>Low risk of bias:</i> The study protocol is available and all of the study's pre-specified (primary and secondary) outcomes that are of interest in the review have been reported in the pre-specified way; the study protocol is not available but it is clear that the published reports include all expected outcomes, including those that were pre-specified (convincing text of this nature may be uncommon) | |
| | <i>High risk of bias:</i> Not all of the study's pre-specified primary outcomes have been reported; one or more primary outcomes is reported using measurements, analysis methods or subsets of the data (e.g. sub scales) that were not pre-specified; one or more reported primary outcomes were not pre-specified (unless clear justification for their reporting is provided, such as an unexpected adverse effect); one or more outcomes of interest in the review are reported incompletely so that they cannot be entered in a meta-analysis; the study report fails to include results for a key outcome that would be expected to have been reported for such a study | |
| | Unclear: Insufficient information to permit judgement | |
| Other bias Bias due to problems not covered elsewhere in the table | <i>Low risk of bias:</i> The study appears to be free of other sources of bias | |
| | <i>High risk of bias:</i> Had a potential source of bias related to the spe- cific study design used; stopped early due to some data-dependent process (including a formal-stopping rule); had extreme baseline imbalance; has been claimed to have been fraudulent; had some other problem | |

Unclear: Insufficient information to assess whether an important risk of bias exists; insufficient rationale or evidence that an identified problem will introduce bias

CONTRIBUTIONS OF AUTHORS

- 1. Draft the protocol: AF, DB, AC
- 2. Study selection: AF, DB
- 3. Extract data from studies: AF, DB
- 4. Enter data into RevMan: AF
- 5. Carry out the analysis: AF, AC
- 6. Interpret the analysis: AF, DB, AC
- 7. Draft the final review: AF, DB, AC
- 8. Disagreement resolution: AC
- 9. Update the review: AF

DECLARATIONS OF INTEREST

- Alicia I Fayad: none known
- Daniel G Buamscha: none known
- Agustín Ciapponi: none known

SOURCES OF SUPPORT

Internal sources

• No sources of support supplied, Other.

External sources

• No sources of support supplied

DIFFERENCES BETWEEN PROTOCOL AND REVIEW

We modified the definitions of intensive and less intensive CRRT according to suggestions of the Editorial Committee and external referees.

'Summary of findings' tables have been incorporated.

INDEX TERMS

Medical Subject Headings (MeSH)

Acute Kidney Injury [mortality; *therapy]; Hypophosphatemia [etiology]; Length of Stay; Randomized Controlled Trials as Topic; Recovery of Function; Renal Dialysis [methods]; Renal Replacement Therapy [adverse effects; *methods]

MeSH check words

Humans